ECONSTOR

Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Erena, Obsa Teferi; Kalko, Mesfin Mala; Debele, Sara Adugna

Article

Technical efficiency, technological progress and productivity growth of large and medium manufacturing industries in Ethiopia: A data envelopment analysis

Cogent Economics & Finance

Provided in Cooperation with:

Taylor & Francis Group

Suggested Citation: Erena, Obsa Teferi; Kalko, Mesfin Mala; Debele, Sara Adugna (2021) : Technical efficiency, technological progress and productivity growth of large and medium manufacturing industries in Ethiopia: A data envelopment analysis, Cogent Economics & Finance, ISSN 2332-2039, Taylor & Francis, Abingdon, Vol. 9, Iss. 1, pp. 1-38, https://doi.org/10.1080/23322039.2021.1997160

This Version is available at: https://hdl.handle.net/10419/270173

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



WWW.ECONSTOR.EU

https://creativecommons.org/licenses/by/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.







Cogent Economics & Finance

ISSN: (Print) (Online) Journal homepage: https://www.tandfonline.com/loi/oaef20

Technical efficiency, technological progress and productivity growth of large and medium manufacturing industries in Ethiopia: A data envelopment analysis

Obsa Teferi Erena, Mesfin Mala Kalko & Sara Adugna Debele |

To cite this article: Obsa Teferi Erena, Mesfin Mala Kalko & Sara Adugna Debele | (2021) Technical efficiency, technological progress and productivity growth of large and medium manufacturing industries in Ethiopia: A data envelopment analysis, Cogent Economics & Finance, 9:1, 1997160, DOI: <u>10.1080/23322039.2021.1997160</u>

To link to this article: <u>https://doi.org/10.1080/23322039.2021.1997160</u>

9	© 2021 The Author(s). This open access article is distributed under a Creative Commons Attribution (CC-BY) 4.0 license.	Published online: 05 Nov 2021.
	Submit your article to this journal 🗹	Article views: 4886
Q	View related articles 🗹	Uiew Crossmark data 🗹
ආ	Citing articles: 5 View citing articles 🗹	





Received: 07 February 2021 Accepted: 20 October 2021

*Corresponding author: Obsa Teferi Erena, College of Business and Economics, Hawassa University, P.O. Box: 05, Hawassa, Ethiopia E-mail: obteferi@gmail.com

Reviewing editor: Christian Nsiah, School of Business, Baldwin Wallace University, Ohio, United States

Additional information is available at the end of the article

GENERAL & APPLIED ECONOMICS | RESEARCH ARTICLE

Technical efficiency, technological progress and productivity growth of large and medium manufacturing industries in Ethiopia: A data envelopment analysis

Obsa Teferi Erena^{1*}, Mesfin Mala Kalko² and Sara Adugna Debele¹

Abstract: The purpose of this study is to assess empirically how the technical efficiency scores for 43 sub-sectors and their determinants over the period 2010 to 2017 show significant variation across the sub-sectors. The study applied a two-step approach for measuring technical efficiency and its determinants. A data envelopment analysis output-orientation (i.e. both CCR & BCC models) is used to estimate technical efficiency scores for 43 sub-sectors over the period 2010 to 2017. Malmquist productivity index (MPI) output orientation is also applied to compute technical efficiency score shows significant variation across the sub-sectors. Thus, we used a Tobit regression model to scrutinize what defines the variation in technical efficiency scores using three years of panel data which covers 2015 to 2017. Moreover, the 43 sub-sectors were further grouped into 14 major sub-

ABOUT THE AUTHOR

Obsa Teferi Erena is an Assistant Professor at the College of Business and Economics at Hawassa University, Ethiopia. His research interest include corporate governance, earning management, knowledge management, technology management, and financial accounting. He has published in Corporate Governance: The International Journal of Business in Society. Email: obteferi@gmail.com Mesfin Mala Kalko is currently a Ph.D. Candidate at the Faculty of Management and Economics at Tomas Bata University in Zlin, Czech Republic. His research interest include corporate governance, corporate social responsibility, corporate finance, innovation, knowledge management, and financial accounting. He has published in Corporate Governance: The International Journal of Business in Society. Email: kalko@utb.cz Sara Adugna Debele is a Lecturer at the College of Business and Economics at Hawassa University, Ethiopia. Her research interest include financial accounting, corporate governance, and corporate social responsibility. She has published in Corporate Governance: The International Journal of Business in Society.

PUBLIC INTEREST STATEMENT

This paper uses data envelopment analysis to measure technical efficiency, technological change, and productivity growth in 43 manufacturing industries over the period 2010 to 2017. Malmquist productivity index (MPI) output orientation is also applied to compute technical efficiency change, technological change, and productivity change. The results show that the sector had experienced a 37 percent technical efficiency in overall average when the CCR model was used. The findings of the study would have implications for policymakers, government, and firm owners in that it provides an insight into the source of productivity growth in the sector.





Email: saraadu2@gmail.com

 \circledast 2021 The Author(s). This open access article is distributed under a Creative Commons Attribution (CC-BY) 4.0 license.

sectors and classified as public and private to examine whether there is a technical efficiency score discrepancy between the same sub-sectors operating under different ownership. For measuring overall technical efficiency, we used two output variables (i.e., value-added and operating surplus) and two input variables (i.e., total fixed assets and a total number of employees). When reducing the sub-sectors to fourteen major groups, the operating surplus was not included, thus we used valueadded and total sales as output variables and total fixed assets, the total number of employees, and cost of raw materials used in the production process as input variables. To shed light on the source of inefficiency, technical efficiency is decomposed into pure technical efficiency and scale efficiency. This study found that the sector had experienced a 37 percent technical efficiency in overall average when the CCR model was used. The study also claims that public owned subsectors are less likely to be efficient than private subsectors. The regression results show the capital expenditure ratio has a significant positive influence on technical efficiency. The Malmquist index result also shows, on average, the sector had registered a 10.5% technological progress and a 13% productivity growth over the period 2010-2017. The findings of the study would have implications for policymakers, government, and firm owners in that it offers an insight into the source of productivity growth in the sector.

Subjects: Economics; Finance; Business, Management and Accounting

Keywords: Technical efficiency; Technological progress; Productivity; DEA; Tobit Model; Ethiopian manufacturing sector

1. Introduction

The manufacturing sector plays a crucial role in production, growth, and job creation (Naudé & Szirmai, 2012). It is the most important engine of long-term growth and development, both in developed and developing countries (McKinsey, 2012). In the former, for instance, in 2015 the manufacturing sector shares 12% and 19% of the gross domestic product (GDP) of the United States and Japan, respectively (UNCTAD, 2015) and it remains a vital source of innovation & competitiveness, making enormous contributions to research & development, exports, and productivity growth (McKinsey, 2012). While, in the latter, the manufacturing sector shares 27% and 16% of the GDP of China and India in 2015, respectively (UNCTAD, 2015) and it continues to provide a pathway from subsistence farming to rising incomes and living standards. For the least developed countries, such as Ethiopia, agriculture makes up the highest proportion of the economy. According to the Ethiopian CSA (2018), the net contribution of the manufacturing sector to the GDP growth rate increased from 0.4% in 2011 to 1.1% in 2017. The agriculture and services sectors accounted for 36.3% and 39.3% in 2017, respectively. This low contribution of the manufacturing sector to the GDP is a common feature of Sub-Saharan African countries. oFr instance, it takes 8.4% of Kenya's GDP (Kenya Association of Manufacturers, 2018) and 2.4% of Djibouti's GDP (United Nations, 2016). In this regard, Ethiopia has shifted its economic strategy from agricultural to industrial lead in 2011. The strategic plan was divided into two five-year plans, with Growth and Transformation Plan I, which covers from 2011 to 2015, and Growth and Transformation Plan II, which covers from 2016 to 2020. Nevertheless, the agriculture sector has continued to dominate the GDP of the country, which was followed by the service sector. Indeed, the manufacturing sector's contribution to GDP demonstrates Ethiopia's infant stage of manufacturing activities or industrialization. Empirical evidence (Ayelign & Singh, 2019; Ogubay, 2015; Hailu & Tanaka, 2015; UNDP, 2017) has also confirmed that the sector had faced several problems, such as limited access to and interrupted electricity power, a low level of export performance and competition, a shortage and irregular supply of domestic raw materials, limited access to and poor quality of internet services, and weak logistic support.

Technical efficiency is the ability of a firm to produce as much output as possible with a specified level of inputs, given the existing technology. It can also be a situation wherein it is impossible, with current technical knowledge, to increase output from given inputs or produce a given output using less than one input without using more of another input (Farrell, 1957). Efficiency is a major problem in Ethiopia. Because Ethiopia'smanufacturing industries were not operating at full capacity (Hailu & Tanaka, 2015), there was a need to improve the sector's efficiency (Bekele & Belay, 2007). Ethiopia has very limited capital but an abundant workforce, and hence its industries are predominantly labor-intensive instead of capital-intensive. Productivity growth might come from the enhancement of productivity based on catching up capability and innovation by effective use of human capital in the labor market and adoption of new technology. Conversely, switching from labor-intensive to capital-intensive would increase productivity if an optimal benefit is achieved from technology change. The Ethiopian manufacturing sector tends to be labor-intensive. This is common in the least developed countries because of the presence of a massive pool of unemployed labor force (Wu, 1993). Being labor-intensive or capital-intensive might not result in efficiency or inefficiency, but being able to produce additional units of production (output) while keeping input constant or reducing input while keeping output constant could lead to the efficient frontier. There have been contradicting results in the literature that suggest capital-intensive firms are more efficient than labor-intensive one. For example, Arrow et al. (1961) suggested that differences in the efficiency of firms arise due to variations in the efficiency of the labor force. Wu (1993), using econometric and group-wise analyses, finds that labor-intensive firms are relatively more efficient than capital-intensive firms. In contrast, Alvarez and Crespi (2003) and Sun et al. (1999) suggest firms that are capital-oriented tend to be more efficient. Similarly, Li and Zhao (2017) indicated that capital-intensive firms were found to perform better and have higher stock value than labor-intensive firms.

There have been few studies on measuring the productivity of manufacturing firms in Ethiopia. For example, Goshu et al. (2017) proposed a framework for measuring productivity in manufacturing companies. Tsegay et al. (2018) applied conventional OLS and panel data to test determinants of performance in manufacturing with regard to the textile and garment industry. Rao and Tesfahunegn (2015) used the Cobb-Douglas production function to examine the performance of manufacturing industries. Abegaz (2013), Hailu and Tanaka (2015), and Ayelign and Singh (2019) estimate the technical efficiency and total productivity changes of medium and large manufacturing firms using a comprehensive panel data set annually collected by the central statistical agency. Bekele and Belay (2007) analyzed technical efficiency and its determinants in the grain mill products manufacturing industry, using the stochastic frontier model. Moreover, the recent study by Oqubay (2018) analyzed the structure and performance of manufacturing industries. Most of the studies stated here have employed a linear function, stochastic frontier approach to computing technical efficiency score which is subject to model diagnostic tests such as the normal distribution of residual terms (which is a part of technical inefficiency score), model identification, and specification. In addition, no attempt has been made in these prior studies to examine what defines the variation in technical efficiency scores of the firms under study. Thus, this study attempts to fill this gap by using two-fold analyses: (1) measuring technical efficiency and total productivity growth using a non-linear programming approach, data envelopment analysis approach, and (2) a Tobit regression model has been employed to analyze determinants of technical efficiency. Furthermore, a comparative analysis has been performed to understand whether the technical efficiency score differs between public-owned industrial groups and privately owned industrial groups.

We assume that this study contributes to the body of knowledge in two ways. First, it has used a more comprehensive analysis to answer the question that addresses why some firms are more efficient than other firms? This question has been partially answered by identifying the potential firm-specific factors that largely define a firm technical efficiency score. Second, the study asserts that public-owned firms are less likely to be efficient than privately-owned firms. The important question to be raised here is: do resource providers worry about their firm's technical efficiency and productivity growth? This question may not be relevant and sound where there are strong share-holders/resource providers' laws and regulatory provisions that impose duties and responsibilities on the management of the company. However, in developing countries such as Ethiopia, share-holders' law and other provisions are scarce and limited in their application, so public-owned firms are assumed to be less efficient than those firms that operate under private investors. The best resolution to this problem would be privatizing public-owned firms. Ethiopia is currently working on a privatization strategy.

The remainder of the paper is organized as follows: Section 2 presents a review of literature relevant to this study. Section 3 presents the methodology employed in the study. Section 4 reports results and discussion and Section 5 presents the conclusion. Sections 6 & 7 present practical implications and limitations and suggestions for further studies, respectively.

2. Literature review

2.1. The link between the relative technical efficiency and technological change to productivity growth

Productivity growth permits a company to increase profit and market share at the micro-level, and it assists a country to create jobs, counteract inflation, and force the necessary industrial restructuring at the macro-level (J.D. Lee & Heshmati, 2009, p.1). There is widespread agreement among academic researchers in the field of growth theory, policymakers, and businessmen that productivity raise is essential for continued economic growth (J.D. Lee & Heshmati, 2009, p.1). In one of the original contributions to economic growth theory, the investigations of economic growth by Abramovitz (1956), Denison (1962), and Kendrick (1956), productivity/efficiency was considered transcendent for clarifying a noteworthy portion of growth rates of physical and labor capital as well as the growth rates of per capita production within the USA. From their conclusions, they asserted that much of the growth was because of productivity or, agreeing to Abramovitz (1956), the measure of our ignorance. Having confirmed the significance of productivity for economic growth, Denison (1962) contended that one of the explanations for its acceleration rested in economies of scale, but this might not be directly influenced.

Among the contributions of Solow (1956) and Swan (1956), who introduced productivity into an economic growth model, where it had been called technical progress. The growth model was supported by the analysis of a neoclassical production function, which assumed constant returns to scale and decreasing returns on inputs. Solow (1956) stated technical progress was an increasing factor of scale by which production was multiplied. Meanwhile, Swan (1956) said technical progress was initially neutral but increased its responsibility for rises in output that were not caused by rises in capital or labor and indirectly increased production by increasing the contribution of capital. In contrast to those models, endogenous models appeared within which technical progress would be internal to the model of economic growth. Among these studies are Lucas (1988), Romer (1986, 1990), who were also known for their attention to increasing incomes at scale and the consideration of models in flawed equilibrium, assuming equilibrium in monopolistic competition and the inclusion of human capital stock in the production function. Though, considering TFP, technical progress, or technological change, there is also the model of Mankiw et al. (1992) which wanted to defend Solow's contributions to economic growth by finding solutions to some of the critiques indicated in the original model. Thus, it was treated as an augmented Solow model with human capital, and to the authors, that alternation better fit the explanation of the growth of nations.

In a similar vein, the mainstream approaches to economies of innovation by Acemoglu and Zilibotti (2001) pointed out that many technologies used by the least developed countries (LDCs) are developed in the advanced economies and are designed to make optimum use of the skills of these richer countries' workforces. Differences in the supply of skills create a mismatch between the requirements of these technologies and the skills of LDC workers and lead to low productivity in the LDCs. Even when all countries have equal access to new technologies, this skill-technology disparity can lead to sizable differences in TFP and production per worker. For example, the evolutionary approach of Nelson and Winter (1973) to the economics of innovation indicated diffusion processes for new technologies and the existence of significant differences among firms in terms of profitability, the technology used, lead to differences in productivity and growth. Similarly, Geroski et al. (1993) in their studies on the profitability of 721 innovating manufacturing firms in the UK, found the number of innovations achieved by manufacturing firms had a positive impact on operating profit. They also indicated innovative firms were more profitable than non-innovative firms in general, although the effect of specific innovation types on firm profit margin was only modest in size.

The term economic efficiency refers to the use of resources to maximize the production of goods and services. In absolute terms, the situation can be called economically efficient if: (1) no one can be made better-off without making someone else worse-off, (2) no additional output can be obtained without increasing the amount of input, and (3) production proceeds at the lowest possible per-unit cost (Sullivan & Sheffrin, 2007 p. 15). Efficiency can be categorized into technical, allocative, or the combination of the two (i.e. total economic efficiency) based on the scope of efficiency targeted (Bhat et al., 2001). Technical efficiency means producing maximum output with given inputs, or equivalently, using minimum inputs to produce a given output (Farrell, 1957). Farrell (1957) considered a production function for a fully efficient firm and analyzed technical efficiency for a production firm as the ratio of the output of any given firm to that of a fully efficient firm. Allocative efficiency deals with the minimizing of the cost of production with a proper combination of inputs for a given level of output and a set of input prices, assuming that the entity examined is working at full technical efficiency. These technical and allocative efficiencies can be combined as a measure of economic efficiency.

TFP can effectively contribute to output growth by improvements in technology and efficiency, as these are two determinants of TFP, under constant returns to scale. If returns to scale are variable, TFP growth can be generated by technical change, efficiency improvement, and scale effects. This also reinforces the potential role played by technical efficiency in determining productivity and therefore the need for the relation of assumptions to accommodate inefficiency and efficiency variations. Technical efficiency reflects firm-specific technical knowledge and effort (Page, 1980), the will, skills, and determination of employees and management (Aigner et al., 1977; L.-F Lee & Tyler, 1978), and the effects of work stoppages, managerial skills, material bottlenecks, worker efforts and other disruptions to production (L.-F Lee & Tyler, 1978).

To explore sources of productivity growth in the presence of inefficiencies, it is essential to appropriately model production technology and inefficiencies among economic agents. Data envelopment analysis (DEA) has been extensively used to analyze productivity growth and inefficiencies. Data envelopment analysis represents a method of analysis that can serve as an aid in identifying best practice performance in the utilization of resources amongst firms of a similar category. Such identification can highlight where the most significant benefits can be made from efficiency improvements and assist organizations to realize their maximum potential. Measurement tools such as DEA are useful in situations where government, subsidies, and a lack of contestability. In these cases, the same old market indicators of performance such as profitability and rates of return cannot be used to measure an organization's economic performance accurately. Despite this, governments and the public at large are still worried that these organizations is performance in the public at large are still worried that these organizations is performance accurately. In these situations, DEA provides comparative monitoring that identifies

variations and hence provides encouragement and direction for the improvement of the performance (Abbott & Doucouliagos, 2003).

Most of the prior literature on productivity focused on input productivity like labor or capital as a measure of input efficiency. A rise in the level of productivity reflects a rise in the efficiency of inputs. Hence, the same level of inputs can produce higher output levels, which suggests a reduction in the cost of production. In other words, it reflects betterment in the input qualities. A study conducted by Bhatia (1990) on misleading growth rates in the manufacturing sector argued that unstable socio-demographic changes and lower levels of technology are causing low productivity in India as compared to the United Kingdom and the United States. In his study of the manufacturing sector in India using data for 21 years (from 1965 to 1985), it was pointed out that factor efficiency was influenced by the factor of production, socio-demographic, socio-politics, development and management of the human resource, workplace, and working condition where a higher capital-labor intensity ratio is associated with a higher level of technology.

2.2. Manufacturing sector of Ethiopia

Ethiopia began its first series of economic reform programs in 1992. The reform programs are aimed at reorienting the economy from a command to a market economy, rationalizing the role of the state, and creating legal, institutional, and policy environments to enhance private-sector investment. Different sectoral policies, strategies, and plans were developed and implemented in an effort to make the manufacturing industry play a great role in the economy. As a result of the economic reforms and priorities given to the sector, its contribution to the economy has increased from 11.4% in 2003/2004 to 13.4% in 2010/11 and within the industry, the construction and manufacturing sub-sectors have registered a high growth rate of 12.8% and 12.1% respectively (MoFED, 2011). The fact that the contribution of the manufacturing sector to GDP is minimal exhibits the infant stage of manufacturing activities or industrialization in Ethiopia. This low contribution of the manufacturing sector to the GDP is the common feature of most developing countries that are especially found in Sub-Saharan African countries. The share of the manufacturing value added (MVA) is one of the indicators which pave the way to assess the sector's performance against other economies.

The Ethiopian manufacturing sector is dominated by food products and beverages and nonmetallic mineral manufacturing sub-sectors. In 2017, the former made up about 26% of the establishments in the manufacturing sector (CSA, 2018). The relatively high number of food products and beverage manufacturing industries is mainly explained by the high local input content and the availability of large local markets for food products and beverages (Befekadu & Berhanu, 2000). In 2017, grain mill products manufacturing firms (GMPMF) contributed about 35% of the manufacturing of food products and beverages industrial group (CSA, 2018). Industries such as metal processing, electrical and electronics, chemical, and other engineering industries, which help build technical capabilities and dynamism, have not yet been developed. Most manufacturing exports are focused on agriculture, including drinks, clothes, shoes, and semi-processed hides. On the other hand, most capital goods and manufactured consumer goods are imported into Ethiopia, which is also heavily reliant on the importation of fuel. On the policy facet, the government is committed to creating a favorable environment for attracting direct foreign investment and promoting domestic investment. A variety of foreign companies from China, India, Turkey, and Japan are presently competing in the country to leverage this opportunity. The preferential dutyfree trade access provided by Ethiopia to the United States of America and European Union markets also provides strategic opportunities (Hailu & Tanaka, 2015).

Ethiopia has abundant resources that can provide valuable inputs for light manufacturing, namely, cattle, which can be used as an input for making leather and leather products; forests, which can be used as an input for the furniture industry; cotton, which can be used as an input for the garment industry; and agricultural land and lakes are used to provide inputs for agroprocessing industries (Dinh et al., 2012). Moreover, Ethiopia has plentiful low-cost labor, which gives it a comparative advantage in less-skilled, labor-intensive sectors (Dinh et al., 2012; Sonobe et al., 2009). In such light manufacturing areas as leather products and apparel, textile, wood products industries, it has a good opportunity for low-cost manufacturing exports.

2.3. Determinants of technical efficiency of manufacturing firms in Ethiopia

The aim of this section is to identify the factors that affect each firm's efficiency levels. These determinants of technical efficiency can be summarized as follows:

2.3.1. Capital expenditure

Prior studies on capital investment in fixed assets indicated a significant and positive relationship between capital expenditures in fixed assets and productivity growth (Abdi, 2008; Delong & Summers, 1991; Gort et al., 1999; Gumbau-Albert & Maudos, 2002; Sala-i-Martin, 1997). For instance, Delong and Summers (1991) found a rising 1% investment share in machinery and equipment could lead to a 0.2 to 0.3% rise in long-run productivity growth. In support of their findings, Sala-i-Martin (1997) pointed out that a 1% increase in equipment investment could lead to a 0.06% rise in output growth, while a 1% rise in non-equipment investment could lead to a 0.06% rise in productivity growth of output. Similarly, Gumbau-Albert and Maudos (2002) indicated that differences in efficiency are typically due to a higher ratio of investment to physical capital if it is believed that new production technologies are integrated into new capital purchases, and that technological improvement accelerates the growth of efficiency/productivity in the sector.

Thus, we propose new capital investment in fixed assets is positively associated with firm efficiency in manufacturing firms in Ethiopia.

2.3.2. Capital intensity

The relationship between capital intensity and technical efficiency has been studied by many scholars with inconsistent results (Latruffe et al., 2004; Mathijs & Vranken, 2000; Sun et al., 1999; Wu, 1993). An important finding of previous studies indicates that capital intensity, measured as capital divided by labor, has a significant and positive impact on technical efficiency in the food, machinery, and electronics sectors of Chinese manufacturing industries (Sun et al., 1999). They pointed out that a rise in the utilization of capital inputs such as machinery and equipment in relation to labor, or capital deepening, is expected to improve productivity and lead to a growth in technical efficiency in these industries. Similarly, Mathijs and Vranken (2000) indicated more capital-intensive farms are efficient in Bulgarian crop farms. In contrast, Latruffe et al. (2004) pointed out more capital-intensive farms are less efficient in crop and livestock farms in Poland. Hence, we propose capital intensity is positively associated with manufacturing firm efficiency in Ethiopia. In prior literature, capital intensity is measured as the ratio of total assets (book value) to the total number of employees (Blomström & Persson, 1983). Abenoja and Lapid (1991) measured capital intensity as the ratio of the gross book value of fixed assets to the total number of production workers. Following Abenoja and Lapid (1991), for this study, we used the ratio of the book value of machinery and equipment of the establishment to the total number of production workers.

2.3.3. Account-book ratio

To our knowledge, extant research has not addressed the impact of internal control on manufacturing firm efficiency. When sound internal controls are maintained and effectively monitored, they are an important aid to enhancing productivity and effectiveness. Firms that maintain proper record-keeping are assumed to be efficient. Firms that keep a complete book of account are in a better position to prudently plan and track the day-to-day operations of their production unit (Bekele & Belay, 2007). This will aid them to improve their technical efficiency level by preventing waste of resources. In this study, the account-book ratio, as a proxy for internal control, is measured as the ratio of firms that maintain books of account to the total number of firms in the industry. Thus, we hypothesize that the account-book ratio is positively related to the technical efficiency of manufacturing firms in Ethiopia.

2.3.4. Skill intensity

Prior literature on the relationship between skill intensity and technical efficiency indicates that skill-intensive firms are more capital-intensive, larger in size, tend to be exporters, and more productive (Bernard & Jensen, 1999). Firms with better managerial skills tend to have higher earnings, production, and technical efficiency (Kirkley et al., 1998). Similarly, Ray (1997) indicates an increase in the proportion of non-production white-collar and managerial staff might impose certain rigidities in the production process, causing slow adjustments to variations in demand. In this study, skill intensity is measured as the ratio of production workers to total employees. Thus, we expect a positive correlation between skill intensity and technical efficiency. This expectation is also consistent with conventional trade theory, which claims that firms with higher skill intensity specialize in higher quality products and tend to be more profitable and efficient (Whang, 2016).

2.3.5. Industry size

Theoretical research on the relationship between firm size and efficiency indicates that larger firms benefit from economies of scale and operate at lower average costs of production, implying that firm size has a positive impact on efficiency. Similarly, a theory developed as a model of firm growth by Jovanovic (1982) indicates larger firms are more efficient than smaller ones. This result is an outcome of a selection process, in which efficient firms grow/prosper and survive, while inefficient firms stagnate or leave the industry. Furthermore, empirical studies on the firm size and efficiency relationship indicate various results. For instance, Lundvall and Battese (2000) indicated technical efficiency increases with firm size. Sun et al. (1999) also pointed out a rise in the size of firms is likely to promote a firm's market share and competitiveness, which in turn is expected to improve a firm's access to new technology, scale efficiency, innovative capability, and productivity. These improvements tend to improve the firm's technical efficiency. Similarly, Sur et al. (2018) indicated a positive association between size and technical efficiency. They found that as the capacity of a firm increases, its efficiency also increases. In contrast, Betancourt and Claque (1975) assumed a negative association between efficiency and firm size. They argue small firms adopt more appropriate technology and foster competitive factors and product markets with their flexibility to respond to changes in technology, product markets, and markets. Hence, we hypothesized that firm size is positively correlated to technical efficiency.

2.3.6. Advertising expense

According to the Resource-Based Theory of the firm, firms that invest in R&D and advertising are more likely to create firm-specific assets that cannot be imitated by their rivals/competitors and serve as the foundation for their long-term competitive advantage. Firms' advertisement and R&D choices, according to Geroski (1995), may help to better capture relative efficiency and its evolution over time. Firms that obtain innovations and conduct advertising improve their efficiency, making them more likely to succeed.

Extant literature on the relationship between advertising expenses and technical efficiency is scarce. OCED (2014) indicated that advertising is an example of firms` responses to competition and is associated with improved productivity. Firms that spend money on advertising tend to be more productive than their competitors. Similarly, advertising expenditures may also be thought of as endogenous sunk costs, as Sutton (1991) suggests, strengthening the firms' perceived brand reputation and increasing customers' willingness to pay for their goods. As a result, advertising is supposed to boost survival chances. Comanor and Wilson (1967) suggest that advertising has an anticompetitive impact because it increases entry barriers, softening the resilience of competition. Following Özçelik and Taymaz (2004), in this study, the advertising ratio is measured as the ratio of advertising expense to total sales. Thus, we hypothesize that advertising expense is positively associated with technical efficiency.

3. Methodology

The manufacturing sector is one of the rapidly growing sectors in Ethiopia. According to the CSA (2018) report, about 3,529 large and medium manufacturing companies are operating in Ethiopia.

About 40% of them are found in Addis Ababa (the capital city of the country), whereas 23% found in the Oromia region, 11% in the Amhara region, and 9% in the Tigray region. Food products and beverages share the highest percentage, at 26%, followed by non-metallics at 18%. The textile industry shares only 1.7%.

3.1. Data and variables

We obtained data from the Ethiopian central statistical agency on large and medium manufacturing industries for the year 2010 to 2017. We focus on this period because it is a period in which the Ethiopian government adopted the Growth and Transformation Plan I (GTP-I) to lay the foundation for a structural transformation in the economy and hoped to transform the country into a lowermiddle-income. As such, the manufacturing sector was expected to play a catalyst role in driving the growth momentum, employing factor inputs, and changing the structure of the exports, which were dominated by agricultural primary commodities.

Forty-three (43) sub-sectors were considered in the analysis, which further classified into 15 major industry aroups. To measure the Malmquist productivity index, eight years of panel data from 2010 to 2017 was used. Technical efficiency is computed for each year. Consistent with Rezitis and Kalantzi (2016) and Lakner et al. (2017), we compute at an aggregate industry level efficiency rather than firm individual level as it is not permitted for data confidentiality. Following prior studies of efficiency (Taymaz & Saatci, 1997; Diewert, 2000; Fu, 2005; Hossain & Karunaratne, 2004; Kim, 2003; Larossi et al., 2009, p.110; Murat & Federica, 2018; Salim & Kalirajan, 1999) this study used two output variables (i.e., value-added at market price and operating profit or surplus) and two input variables, namely, capital and labor. For instance, Hossain and Karunaratne (2004), Kim (2003), Larossi et al. (2009), Murat and Federica (2018), and Salim and Kalirajan (1999), employed value-added output variable. While Tzeng and Huang (2013) utilized operating surplus as an output measure. In this study, value-added is measured as the gross value of production less industrial and non-industrial costs. Similarly, operating surplus or profit is measured by deducting the total cost from total revenues (Tzeng & Huang, 2013). Since using a large number of input variables can cause most or all of the firms to be ranked as efficient (Leibenstein & Maital, 1992), we included capital and labor as inputs to minimize potential measurement errors. Various capital input measurements have been used in several empirical studies. For instance, Kim (2003) measured capital by the number of tangible fixed assets for Korean manufacturing firms. Hossain and Karunaratne (2004) also measured capital as the gross fixed assets. In this study, following the existing literature, capital is measured by gross fixed assets. There have been various ways to measure labor in literature. Taymaz and Saatci (1997) measured labor as the total number of hours worked in production. Diewert (2000) measured labor as total wages and salaries paid. Fu (2005) measured labor by the number of employees. In this study, labor is measured by the number of employees (both permanent and seasonal employees). Furthermore, the original 43 sub-sectors were classified into fourteen (14) major sub-sectors and each sub-sector is separated as public-owned and private-owned to examine whether the technical efficiency score varies between the public and private-owned industries. As we could not get data on operating surplus at the major industry level, value-added and total sales were assumed as output variables. The input variables used for this particular analysis include fixed assets, the number of employees, and the costs of raw materials.

3.2. Model specification

To measure technical efficiency, various approaches have been addressed in the literature over the last four decades. The most extensively applied methods are DEA and Stochastic Frontier. The first DEA model was proposed by Charnes et al. (1978) and was later named the CCR model. DEA is an approach to measuring the relative efficiency of a set of decision-making units (DMUs) with multiple inputs and multiple outputs using mathematical programming. DEA has relied on certain simple assumptions like positive numerical data should be available for all output and input items of DMUs. The input and output items are easily disposed of by management. Unlike a production function, DEA does not assume the functional relationship between variables (input and output)

and insensitive to model specification (Alvarez & Crespi, 2003). However, the DEA approach has been criticized in literature for two reasons (Green & Mayes, 1991). First, all firms may not have access to the same technology. If this is the case, there is no basis for measuring technical efficiency relative to a single frontier, since each firm may be efficient with regard to its own set of production possibilities. Second, the measure neglects differences in the physical environment and product differentiation of firms. If the environment varies between firms, apparent differences in efficiency may arise from the degree to which the environment of a particular firm is favorable or unfavorable. These two problems were minimized in this paper by considering industry-level analysis rather than individual firms.

CCR model assumes an input-oriented approach with constant return to scale (CRS). The inputoriented approach reflects minimizing input quantity while keeping the current output constant (Farrell, 1957). In contrast to the CRS assumption, an alternative model called a variable return to scale (VRS) has been developed by Banker et al. (1984). The ideal way to introduce DEA is via the ratio form. For each firm, we would like to obtain a measure of the ratio of all outputs over all inputs (Coelli, 1996).

Say we have a population of *n* productive units, DMU1, DMU2, ..., DMUn.

Each unit produces s outputs while consuming m inputs. Let us write an input matrix as follows:

X = [xij, i = 1, 2, ..., m, j = 1, 2, ..., n] and an output matrix

Y = [yij, i = 1, 2, ..., s, j = 1, 2, ..., n]. The q-th line—i.e.

Xq and Yq—of these matrixes show quantified inputs/outputs of unit DMUq. The efficiency rate of such a unit can then be generally expressed as:

Weighetedsumofoutputs	$\sum_{j=1}^{s} u_i y_i q$	(1	•
Weightedsumofinputs	$=\overline{\sum_{i=1}^{m} v_i x_i q}$	(1	.)

Where:

vj, j = 1, 2, ..., m, are weights assigned to j-th input,

ui, i = 1, 2, ..., s, are weights assigned to i-th output

DEA models derive input and output weights by means of an optimizing calculation. Based on that, DMU can be classified as efficient and inefficient. Inefficient units tell us target values of inputs and outputs which would lead to efficiency. In other words, the actual output obtained is less than the maximum capacity that the firm can produce, so the inefficiency score is the difference between the actual outputs produced and the firm's production capacity.

3.3. CCR and BCC Models

Following empirical studies, we adopted CCR and BCC models to estimate the relative technical efficiency and the Malmquist productivity index in order to measure total productivity change (TFP). Sarkis (2000) argued that the use of the CCR and BCC models together help to determine the overall technical and scale efficiencies of the firm and whether the data exhibit variable returns to scale.

To estimate efficiency scores and identify sources of inefficiency, this study applied outputoriented DEA because we assume that manufacturing industries are more likely to influence their output level than the input. The sector is suffering from a lack of adequate raw materials, skilled labor, less access to machinery and equipment due to foreign currency problems. The sector, therefore, given the resources available to it, can take maximum effort to obtain the optimal output. This seems the sound strategic approach in the current real Ethiopian environment.

First, we have adopted the CCR model which assumes Constant Return to Scale (CRS) according to the following equation (Charnes et al., 1978):

$$\min\theta - \varepsilon \left(\sum_{i=1}^{m} \mathbf{S}_{i}^{-} - \sum_{r=1}^{s} \mathbf{S}_{r}^{+} \right)$$
(2)

Subject to:

$$\sum_{i=1}^{n} \mu_{i} \mathbf{x}_{i} + \mathbf{S}_{i}^{-} = \theta_{k} \mathbf{x}_{k} \mathbf{i}_{k} \mathbf{i} = 1, \dots, \mathbf{m}$$
(3)

The scalar variable is the proportional reduction which should be applied to all inputs of DMU, in order to make them efficient, and ε is non-Archimedean defined to be smaller than any positive real number, whereas S⁻ and S⁺ are "slack variables" and a standard linear programming terminology for additional variables added to the model in order to convert inequality constraints to equality constraints. This terminology in DEA is also used when additional improvement is possible in specific inputs or outputs.

Coelli (1996) indicated that the use of the CRS specification when some of the DMUs are not running at optimal scale will result in measures of technical efficiency which are mixed up with scale efficiency. The assumption of CRS that all DMU are operating at optimal scale is quite unrealistic. It might hold true where there are no market inconsistencies or fluctuations, like no competition, no technological constraint, and changes in customer behavior which are seldom in the real environment. To overcome this problem, Banker et al. (1984) suggested their model known as the BCC model. It improves the CCR model by introducing a variable that represents the returns to scale. The BCC model allows a calculation of technical efficiency that is free from the scale efficiency effects. In the BCC model, the formulation is written as below (Banker et al., 1984)

$$\min\theta - \varepsilon \left(\sum_{i=1}^{m} \mathbf{S}_{i}^{-} - \sum_{r=1}^{s} \mathbf{S}_{r}^{+} \right)$$
(4)

Subject to

 $\sum_{i=1}^{n} \mu_i x_i + S_i^- = \theta_k x_{ik} i = 1, \dots, m$

$$\sum_{i=1}^{n} \mu_{i} y_{rj} - S_{r}^{+} = y_{rk} r = 1, \ldots, s$$

$$\sum_{j=1}^{n} \mu_{j=1} j = 1, \dots, n$$

$$\mu_{j}, S^{+}, S^{-} \ge 0, j = 1, \dots, n$$
 (5)

A DMU is BCC-efficient if and only if $\theta^* = 1$ and all slacks are zero. The envelopment surface in BCC mode1 is variable returns to scale and this is the result of the presence of the convexity constraint $(\sum_{i=1}^{n} \mu j = 1)$.

3.4. Determinants of efficiency

In the overall technical efficiency analysis, both under CRS and VRS, there is a high variation of efficiency scores among the sub-sectors. The variation in efficiency scores is not surprising because the sub-sectors under study comprise all medium and large-scale manufacturing firms operating in different regions of the country. Accordingly, there is heterogeneity between sub-sectors, for instance, in terms of age, geographical location, capital, skilled manpower, and access to markets. Given these factors, we believe that firm-level data is required to conduct a more detailed analysis to understand the variation in efficiency. In this study, based on the data obtained, we have tried to regress efficiency against industry characteristics, in part to answer the question: why are some sub-sectors more efficient than others? The total technical efficiency estimated by CRS is taken as a dependent variable. Since the dependent variable, technical efficiency takes a value between 0 and 1, which are bounded or censored, using ordinary least squares would provide biased estimates as noted in Martin and Page (1983). Thus, the Censored Tobit maximum likelihood method is applied to estimate parameters. Based on the data obtained, the following determinant factors were taken to define the variation in efficiency, namely, advertising ratio, skill intensity, industry size, capital expenditure ratio, account book ratio, and capital intensity. There are several factors recommended in the empirical literature (Alvarez & Crespi, 2003) like export performance, firm owner education level, import performance, research and development, but we could not include these variables in the model due to data unavailability. The Tobit model is expressed as follows:

$$Efficiency_{i,t} = \alpha_{i,t} + \beta_1 A R_{i,t} + \beta_2 A B R_{i,t} + \beta_3 S I_{i,t} + \beta_4 I S_{i,t} + \beta_5 C E R_{i,t} + \beta_6 C I_{i,t} + \varepsilon_{i,t}$$
(6)

Where: *Efficiency*_{*i*,*t*} denotes technical efficiency of the ith sub-sector at time t, AR represents advertising ratio and is measured as the ratio of advertising expense to total expenses, ABR is account book ratio measured as firms which have a book of account to a total number of firms in the sub-sector. SI represents skill intensity measured as production workers divided by total employees, IS denotes industry size measured as firms with 50 and over employees divided by the number of firms in the sub-sector. CER is the capital expenditure ratio as measured by annual capital expenditure divided by total fixed assets. CI represents capital intensity as measured by the book value of machinery and equipment to production workers and $\varepsilon_{i,t}$ is the error term.

3.5. Malmquist productivity index

The most popular approach for productivity change (growth) measurement in DEA is the Malmquist productivity index (MPI), which was named after Professor Sten Malmquist, on whose ideas the MPI is based and was designed by Caves et al. (1982). It is an index representing the TFP growth of a DMU which is considered as an entity transforming inputs into outputs. It reflects progress or regress, along with progress or regress in frontier technology between two periods of time. In a non-parametric approach, productivity growth is the product of catch-up and frontiershift terms. The catch-up (or recovery) term relates to the degree to which a DMU improves or worsens its efficiency. The frontier-shift (or innovation) term reflects the change in the efficient frontiers between the two time periods. (Catch-up) > 1 indicates progress in relative efficiency from period 1 to 2. While (Catch-up) = 1 and (Catch-up) < 1 indicate no change and regress in the efficiency, respectively (W.W. Cooper et al., 2004). The frontier shift is defined by geometric mean. Frontier-shift > 1 indicates progress in the frontier technology around DMU from period 1 to 2, while (Frontier-shift) = 1 and (Frontier-shift) < 1 indicate the status guo and regress in the frontier technology, respectively. Therefore, the Malmquist index is the result of (Catch-up) X (Frontiershift), in that MI > 1 indicates progress in the TFP of the DMUo from period 1 to 2, while MI = 1 and MI < 1 respectively indicate the status quo and deterioration in the TFP (Färe et al., 1994).

The Malmquist index measure of productivity change decomposed into technological progress (technical change) and technical efficiency change (pure technical efficiency change and scale efficiency change). Technological change means that the set of feasible combinations of input and output quantities expands or the curve shifts in the productivity limit, while technical efficiency

change means that the firm moves closer to or further away from the frontier or changes in efficiency reaching the production limit (Balk, 2001). These two components are independent of each other. There can be technological change without efficiency change, and efficiency changes without technological change. Consider a single firm, going from a base period input-output quantity combination (x° , y°) to a comparison period combination (x^1 , y^1). The single-input single-output case's productivity change is then measured by the ratio:

$$\frac{y^1/y^0}{x^1/x^0} = \frac{y^1/x^1}{y^0/x^0}$$
(7)

It indicates that the ratio of output over the input quantity index number or as the ratio of output quantity per unit of input. In the multiple-input multiple-output case productivity function measured $F(x^1, y^1, x^o, y^0)$

$$F(\lambda x^{0}, \mu y^{0}, x^{0}, y^{0}) = \frac{\mu}{2}(\lambda, \mu > 0)$$
(8)

MPI comprises pure technical efficiency change, scale efficiency change, and technological change.

Technological change

$$TC_0^{1,0}(x,y) = \frac{D_0^1(x,y)}{D_0^1(x,y)^2}$$
(9)

Efficiency change

$$EC_0(x^1, y^1, x^0, y^0) = \frac{D_0^1(x^1, y^1)}{D_0^0(x^0, y^0)^2}$$
(10)

4. Results and discussion

This section presents technical efficiency (TE), pure technical efficiency (PTE), and scale efficiency (SE) values of 43 large and medium manufacturing subsectors in the period 2010–2017. The same data were used to compute technological change and total factor productivity (TFP) change during the study period.

4.1. Descriptive statistics

Descriptive statistics provide a guarantee of whether or no variation in the data is large. Table 1 reports the descriptive statistics (mean, standard deviation, minimum and maximum) for output and input variables used in the study. The mean value of the value-added, output variable is Birr 1,103,119, the minimum value is Birr 557, 272 and the maximum value is Birr 13,512,495. Whereas the mean value of operating surplus/profit, the output variable is Birr 696,044.9. The minimum and maximum values are Birr 1,024,703 and Birr 8,263,285, respectively. The sampled sub-sectors, on average, had 6,038 employees (input1) over the observation years. The minimum and the maximum number of employees were 4 and 80,016, respectively. Fixed assets, input variable had Birr 1,230,444 mean value with Birr 41 minimum & Birr 52,373,626 maximum values.

On the other hand, the value of the standard deviation of all variables exceeds its corresponding mean, showing there has been big variation among sub-sectors in the period 2010 to 2017. One of the reasons for the variation could be the size and technological differences. Since the model used in this study is non-parametric so that data variation does not have an impact on computing technical efficiency score.

4.2. Results of data envelopment analysis models

This section discusses the results obtained from the DEA models: the CCR model with the assumption of CRS and the BCC model under the assumption of VRS. The study employed the outputoriented approach, assuming that maximizing output while keeping input constant is more feasible than minimizing the use of the input quantity without reducing the current level of output in Ethiopian manufacturing firms. It is also the Ethiopian government's strategy to maximize the production of manufacturing industries in order to build industry lead economies. The technical efficiency obtained from a CRS is decomposed into two components: pure technical inefficiency and scale inefficiency. Technical efficiency is computed by a CCR model, whereas a BCC model estimates pure technical efficiency. If there is a difference between the two efficiency values, there is scale inefficiency which is computed from the ratio of technical efficiency to pure technical efficiency (TE/PTE).

This study measured relative efficiency by comparing the best practice sub-sector in the sample. The source of inefficiency can be computed by comparing the relative sizes of various efficiency measures. If pure technical efficiency is greater than scale efficiency, then inefficiency is caused by scale inefficiency (Banker et al., 1984).

4.3. CCR model results

The CCR model measures the overall or technical efficiency under the assumption of CRS. CRS assumes that a change in the input changes the output proportionately. Table 2 presents the overall efficiency of the sampled sub-sectors.

As it can be observed in Table 2, in 2010, four sub-sectors were relatively technically efficient, namely sugar and sugar confectionery; tobacco; other chemical products; and cement, lime and plaster. These sub-sectors had relatively efficiently utilized their inputs in the year. It indicates the management's success in allocating and converting inputs to outputs. On average, the sector achieved 34.3 percent technical efficiency, implying that if best practices had been implemented, the sector could have increased output from the same input quantity by 65.7 percent. The number of relatively technical efficient sub-sectors declined to three in 2011. Malt liquors and malt; paints, varnishes and mastics; and basic iron and steel were efficient. Out of the inefficient subsectors (40 subsectors), knitting mills from the textile major sector scored the least efficiency value in the year. This indicates poor use of resources and inappropriate selection of inputs to produce output. The sample sub-sector efficiency value rose by 0.1 percent on average (from 0.343 in 2010 to 0.344 in 2011).

A larger number of technically efficient sub-sectors were found in 2012. Namely, macaroni and spaghetti; tobacco; wood and product of wood; paints, varnishes and mastics; cement, lime and plaster; other fabricated metal; and parts & accessories for the vehicle were efficient. The tobacco product sub-sector has registered better efficiency scores over three consecutive years (2010, 2011 & 2012). The inefficient sub-sectors can further be decomposed into two: sub-sectors that have efficiency scores more than the average (0.507 for 2012) and less than the average. Nine sub-

Table 1. Descri	iptive statistics				
Variable	Obs.	Mean	SD	Min	Max
Value added	344	1,103,119	1,722,435	557,272	13,512,495
Operating profit	344	696,044.9	1,194,240	1,024,703	8,263,285
Number of employees	344	6,038	9,564	4	80,016
Fixed assets	344	1,230,444	3,575,366	41	52,373,626

Note: the data value is in ('000) and in Ethiopian Birr; obs. 344 (43*8).

sectors have efficiency scores more than the average, while twenty-seven (27) sub-sectors are found with an efficiency score less than the average. The overall average efficiency scores of the sector have increased from 0.344 in 2011 to 0.507 in 2012 as some sub-sectors have shown improvement in resource utilization to produce products. Therefore, by adopting best practices, the sector could have on average produced more products by 49.3% than actually produced from the current level of inputs quantity.

The number of efficient sub-sectors slightly declined from 2012 to 2015. Three sub-sectors were efficient in 2013, whereas two sub-sectors were in 2014 and 2015. The average efficiency scores also declined from 2012 to 2016. Moreover, the average technical efficiency per sub-sector over the observation years (2010 to 2017) ranges from 0.122 for spinning, weaving & finishing of the textile to 0.916 for tobacco products. This implies there is a high-efficiency variance among the sampled sub-sectors over the observation years. This result is consistent with some prior studies such as Hailu and Tanaka (2015) who find there is technical efficiency score variation across Ethiopian manufacturing firms, suggesting shortage of raw materials supply was the main cause of the relative technical inefficiency of the sector.

On overall average, the sector understudy had a 37% technical efficiency value from 2010 through 2017. Our result is very consistent with the recent study by Ayelign and Singh (2019), who found that in the overall average Ethiopian medium and large-scale manufacturing industries registered 36.7% technical efficiency over the period 1996-2015. Low technical efficiency and productivity seem to be a common problem in manufacturing industries in Sub-Saharan African countries. For instance, the Kenyan manufacturing sector had low overall productivity and large productivity differences across industries (World Bank, 2014). Diaz and Sanchez (2007) indicated consistent findings that most industries in the manufacturing sector were inefficient and the inefficiency was greater among large firms than small firms. Abegaz (2013) addresses that Ethiopian manufacturing industries have the capability to use imported technology, but improvement and adoption of the technology are weak. This means that the sector has not utilized its maximum capacity. UNCTAD (2015) also provides evidence that the lack of access to and sharing of R&D facilities continues to hinder the ability of local firms to take advantage of opportunities both within Ethiopia and in other emerging markets. The World Economic Forum's Global Competitiveness Index (GCI) ranks Ethiopia 109th out of 140 countries with a score of 3.7 out of 7.0 in the 2015-16 report. It reflects that Ethiopian firms are not as competitive in the international market because innovative activity in the industry is very low. In conjunction with this data, the research and development (R&D) share of GDP was 0.5% in 2015 (UNCTAD, 2015). The aforementioned factors would collectively affect the efficiency of the sector.

4.4. BCC model results

The BCC model evaluates whether increasing, decreasing, or constant returns to scale would be taken to improve the efficiency value found. CRS arises when a percentage increase in (all) inputs produces the same percentage increase in outputs. However, VRS occurs when a proportionate increase in inputs produces a smaller (larger) proportionate increase in outputs (W. W Cooper et al., 2006, p.125). This assumption of the BCC model decomposed into decreasing returns to scale and increasing return to scale. In a decreasing return to scale, an increase in input produces a smaller increase in output. An increasing return to scale arises when an increase in inputs yields a larger increase in output. A VRS model allows the best practice level of outputs to inputs to vary with the size of industries and it also measures the efficiency of the management in utilizing inputs that are free of scale efficiency.

The CRS technical efficiency of the sample industries is divided into pure technical efficiency (PTE) and scale efficiency (SE) in the BCC model. PTE measures the extent to which an industry can increase its output (in fixed proportion) while remaining within the VRS frontier. Thus, technical efficiency measures the industry's overall success in maximizing its output. SE reflects the extent to which an industry projected to the VRS efficiency frontier can further increase its output (again

I able 2. lechn	ical erriciency of	r sub-sectors con	uputea by LLK	model under LKA	o assumption				
Sub-sectors	2010	2011	2012	2013	2014	2015	2016	2017	Average
Processing & preserving of meat, fruit & vegetables	0.27	0.094	0.163	0.114	0.276	0.171	0.054	0.151	0.16
Vegetable & animal oils and fats	0.168	0.16	0.272	0.311	0.348	0.399	0.692	0.803	0.39
Dairy	0.259	0.31	0.285	0.499	0.39	0.355	0.326	0.486	0.36
Grain mill	0.118	0.122	0.194	0.185	0.172	0.341	0.141	0.232	0.188
Animal feeds	0.237	0.137	0.36	1	0.12	0.29	0.358	0.788	0.41
Bakery	0.113	0.276	0.839	0.154	0.145	0.159	0.145	0.135	0.24
Sugar and sugar confectionery	1	0.68	0.35	0.415	0.444	1	0.538	0.517	0.618
Macaroni & spaghetti	0.259	0.237	1	0.161	0.19	0.071	0.095	0.198	0.27
Food products n. e.c.	0.179	0.163	0.262	0.377	0.464	0.279	0.319	0.162	0.27
Distilling, rectifying & spirits	0.794	0.682	0.682	0.255	0.288	0.293	0.147	0.31	0.431
Wines	0.745	0.884	1	0.854	0.797	0.45	0.562	0.771	0.757
Malt liquors & malt	0.391	1	0.88	0.843	0.464	0.197	0.781	0.677	0.65
Soft drinks & mineral water	0.455	0.6	0.226	0.198	0.315	0.59	0.135	0.119	0.32
Tobacco products	1	0.685	1	1	1	0.648	1	1	0.916
Spinning, weaving & textiles	0.133	0.079	0.287	0.182	0.03	60.0	0.071	0.11	0.122
									(Continued)

Table2. (Continu	ed)								
Sub-sectors	2010	2011	2012	2013	2014	2015	2016	2017	Average
Cordage, rope, twine & netting	0.155	0.358	0.218	0.08	0.123	0.242	0.171	0.086	0.179
Knitting mills	0.02	0.018	0.032	0.317	0.27	0.03	0.161	0.333	0.147
Wearing apparel except fur apparel	0.084	0.214	0.424	0.122	0.195	0.162	0.517	Ţ	0.339
Tanning & dressing of leather, luggage & handbags	0.11	0.258	0.32	0.484	0.299	0.142	0.195	0.898	0.33
Footwear	0.12	0.194	0.809	0.148	0.265	0.316	0.127	0.161	0.26
Wood & wood products	0.035	0.083	1	0.785	0.041	0.153	0.192	0.181	0.308
Paper	0.251	0.281	0.19	0.078	0.143	0.243	0.33	0.442	0.244
Publishing & printing services	0.261	0.226	0.245	0.383	0.227	0.342	0.482	0.225	0.298
Basic chemicals	0.264	0.182	0.376	0.419	0.308	0.289	0.491	0.111	0.3
Paints, varnishes & mastics	0.889	-	1	0.771	0.897	0.907	0.309	0.607	0.79
Pharmaceuticals & medicinal chemicals	0.305	0.682	0.435	0.536	0.595	0.172	0.425	0.175	0.415
Soap and detergents cleaning	0.332	0.228	0.771	0.469	0.263	0.371	0.239	0.218	0.361
Chemical products n.e.c.	1	0.077	0.374	0.38	1	0.258	0.307	0.642	0.504
Rubber	0.14	0.182	0.577	0.169	0.353	0.366	1	0.894	0.46
Plastic	0.282	0.224	0.258	0.321	0.281	0.185	0.261	0.216	0.25
									(Continued)

Table2. (Contin	ued)								
Sub-sectors	2010	2011	2012	2013	2014	2015	2016	2017	Average
Glass	0.21	0.23	0.547	0.364	0.374	0.623	0.226	0.309	0.36
Structural clay	0.104	0.213	0.183	0.101	0.191	0.132	0.084	0.182	0.14
Cement, lime & plaster	1	0.862	1	0.324	0.946	0.496	0.23	0.242	0.63
Articles of concrete & plaster	0.096	0.129	0.251	0.185	0.315	0.479	0.213	0.277	0.24
Non-metallic mineral	0.089	0.132	0.34	0.314	0.098	0.151	0.083	0.143	0.168
Basic iron & steel	0.278	0.294	0.509	0.377	0.344	0.417	0.291	0.321	0.35
Structural metal	0.327	0.142	0.409	0.208	0.141	0.414	0.174	0.284	0.262
Cutlery, hand tools & general hardware	0.358	0.288	0.69	0.057	0.381	0.35	0.153	0.218	0.31
Other fabricated metal	0.395	1	1	0.554	0.394	0.622	0.225	0.028	0.52
Other general- purpose machinery	0.281	0.255	0.158	0.165	0.405	0.518	0.269	0.289	0.29
Parts & accessories for motor vehicles	0.658	0.361	-	0.692	0.457	1	0.273	0.344	0.59
Passenger cars, commercial vehicles & busses	0.26	0.375	0.491	-	0.558	0.042	-	0.437	0.52
Furniture	0.308	0.176	0.386	0.224	0.238	0.156	0.197	0.302	0.24
Mean	0.343	0.344	0.507	0.385	0.361	0.347	0.325	0.373	0.37

in fixed proportions) while remaining within the CRS frontier. Thus, SE measures the extent to which a firm can increase output by moving to a part of the frontier with more beneficial returns to scale characteristics.

The decomposition is needed to identify the sources of inefficiency by comparing the PTE and SE. When PTE exceeds SE, the source of inefficiency is due to scale inefficiency (inappropriate selection of scale size). In other words, if there is a difference between the technical efficiency score (CRS technical efficiency and VRS PTE), then it demonstrates scale inefficiency. Conversely, if SE is higher than PTE, then the source of inefficiency is due to poor utilization of inputs, i.e., pure technical inefficiency.

Table 3 reports results obtained from BCC model VRS. The results entail CRS technical efficiency, VRS technical efficiency, and scale efficiency. We present the results of the most recent three years observations in the data set, 2015–2017 for clarity.

In 2015, the number of technically efficient sub-sectors was two (2), about 5% of the sample when VRS TE was assumed, and five (5), 11.6% when CRS TE was assumed. Three sub-sectors found with scale efficiency, suggesting an appropriate selection of inputs and operating on the most productive scale size. Out of the inefficient sub-sectors, 38 sub-sectors experienced poor utilization of inputs as the source of inefficiency is pure technical inefficiency. However, two subsectors had higher PTE than SE which implies that the source of inefficiency is scale inefficiency. This indicates the industries are operating at an inappropriate scale. When looking at the type of scale, twenty-seven (27) sub-sectors (e.g., dairy products; bakery products; footwears; rubber products) appeared to have an increasing return to scale, showing that a proportionate increase in inputs yields a larger proportionate in the outputs. These industries would improve their efficiency by expanding the scale of operation. In contrast, eleven (11) industries (e.g., furniture; soap & detergent cleaning; plastic products; cement, lime & plaster) experienced a decreasing return to scale i.e., a proportionate increase in inputs produces a lower proportionate increase in outputs. This implies the industries have supra-optimal scale size (i.e. operates at the rising portion of long-run average cost curve) and thus, downscaling is needed for achieving efficiency frontier. Five sub-sectors, bakery products, sugar and sugar confectionery, macaroni and spaghetti, tanning and dressing of leather, and parties and accessories for a motor vehicle operate at a flatter portion of the long-run average cost curve that means a constant return to scale.

About 11.6% of the sub-sectors have experienced PTE as computed by VRS in 2016. Three subsectors (tobacco products, rubber products, and passenger cars, commercial vehicles & busses) appeared efficient both by CRS TE and VRS TE. All of the scale-inefficient sub-sectors experienced a decreasing return to scale, i.e., an increase in proportionate usage of input produces the less proportionate increase in outputs. On average, the sample sub-sectors recorded a 0.531 of PTE and a 0.629 of SE suggesting the source of inefficient are pure technical inefficient. Generally, the sector was at poor resource management and converting inputs to the output.

When looking in 2017, four sub-sectors, namely soft drink and mineral water, tobacco products, wearing apparel, and passenger cars, commercial vehicles and busses observed as pure technical efficiency as measured by VRS whereas, three sub-sectors (footwears, tobacco products, and soft drink and mineral water) were scale efficient. Out of the inefficient sub-sectors, 20 sub-sectors experienced a decreasing return to scale and 16 sub-sectors experienced an increasing return to scale.

On average, pure technical efficiency scores declined in 2017 indicating some sub-sectors had used excess inputs to produce output as compared to in 2016. However, a scale efficient score on average shows improvement in the year. The mean of scale efficiency was higher than the mean of pure technical efficiency. It reveals the source of technical inefficiencies of the sector is pure technical inefficient. The industries need to properly manage their input utilization in order to be efficient.

On average, the industries had faced a 49% of pure technical inefficiency over the observation periods. It shows a firm's inability to exploit inputs due to the poor skills of both operatives and management. The overall scale efficiency of 76% shows an appropriate section of production scale by the industries. It implies the organizational source of inefficiency. The Ethiopian manufacturing sector is characterized by a cheap labor force. This would benefit the sector in minimizing production cost as wage/salary is low. It could also be detrimental to the sector because of the inappropriate utilization of the resources by unskilled, low-cost labor forces. In essence, cheap labor implies less-skilled labor or labor incentive. It can be assumed that being labor or capital-intensive would not yield a guarantee for efficiency, but the quality of labor or capital utilized could determine the efficiency of a firm. In other words, a firm can be efficient, if it's able to gain the optimum benefit from the resources utilized such as labor, capital, raw materials, or electric power.

When looking at the annual average efficiency score in each observation period, the lowest average TE score was reported in 2016. The highest average TE and PTE were observed in 2012. The SE with the highest efficiency score was observed in 2010. The sample sub-sectors had shown higher SE in all observation periods and in the overall average.

We further classify the sub-sectors considered in the previous analysis into fourteen major subsectors under both government and private ownership in order to measure technical efficiency and compare whether public sub-sectors are more efficient or not than private sub-sectors. To do the analysis, 2015 and 2017 observations were taken, assuming the sub-sectors are operating in the same environment and market. The results presented in Table 4 and 5 indicates, on average, private-owned sub-sectors had registered better total technical efficiency (0.821), PTE (0.939) and SE (0.876) than public-owned sub-sectors and they are relatively efficient. When comparing subsector to sub-sector, public-owned food products and beverages and machinery and equipment are efficient both under CRS and VRS models, whereas the same sub-sectors under private ownership are inefficient. On average, the technical inefficiency of public-owned sub-sectors highly driven by pure technical inefficiency suggesting they had poorly managed usage of resources. In contrast, scale inefficiency contributes more to the total inefficiency of private sub-sectors. It implies private-owned sub-sectors showed an inappropriate combination of resource use in 2015.

In 2017, the efficiency score has declined for both public and private sub-sectors in terms of total technical efficiency (CRS), PTE & SE (VRS). It is noted that the mean efficiency scores of private sub-sectors are much higher than public-owned sub-sectors, indicating private sub-sectors better manage resources and appropriately mix inputs to obtain the maximum benefit. Indeed, the number of private firms in each sub-sector is quite larger than the number of public firms in the same sub-sector. One of the reasons for the imbalance number of firms is privatization. The government sells state-owned firms to private investors that cause a decline in public firms and increase private firms at the same time. Given this practical issue, we assume the variation between the efficiency of the private and public firms might partially be defined by size (in terms of total asset proxy or number of employees proxy). The result is consistent with Alvarez and Crespi (2003) and Gumbau-Albert and Maudos (2002) findings that public firms on average tend to be less efficient as compared to private firms.

4.5. Determinants of technical efficiency: Tobit regression results

The important finding observed in this study is the capital expenditure ratio has a positive effect on technical efficiency, suggesting a significant investment in new capital would lead the firm to an efficiency level. A firm with advanced machinery and equipment more likely to engage in innovative products which in turn increases production and sales performance when minimizing the length of production time, inventory and accounts receivable turn over and other irrelevant costs. On the other hand, a higher depreciation and maintenance cost would offset the advantage of

I able 3. le	cunical erricie	ency (IE), pu	ire tecnnica	remotency (Pu	EJ, SCOIE EI	liciency (SE),	unu type o	I scale (ILS)				
Sub-		20	15			201	6			201	7	
sectors	2	PTE	SE	TFS	31	PTE	SE	TFS	ΤE	PTE	SE	TFS
Processing & preserving of meat, fruit & vegetables	0.171	0.174	0.983	irs	0.054	0.059	0.913	drs	0.151	0.198	0.762	drs
Vegetable & animal oils and fats	0.399	0.419	0.95	is	0.692	0.757	0.915	drs	0.803	0.808	0.993	irs
Dairy	0.355	0.363	0.977	irs	0.326	0.357	0.914	drs	0.486	0.491	0.989	irs
Grain mill	0.341	0.465	0.733	drs	0.141	0.477	0.295	drs	0.232	0.375	0.617	drs
Animal feeds	0.29	0.335	0.866	irs	0.358	0.468	0.764	drs	0.788	0.869	0.907	drs
Bakery	0.159	0.159	1	1	0.145	0.326	0.446	drs	0.135	0.184	0.734	drs
Sugar and sugar confec tionery	1	1	1	1	0.538	1	0.538	drs	0.517	0.518	0.999	
Macaroni & spaghetti	0.071	0.071	0.994	1	0.095	0.22	0.431	drs	0.198	0.277	0.713	drs
Food products n. e.c.	0.279	0.281	0.994	irs	0.319	0.386	0.827	drs	0.162	0.303	0.535	drs
Distilling, rectifying & spirits	0.293	0.294	0.994	irs	0.147	0.352	0.417	drs	0.31	0.346	0.896	drs
Wines	0.45	0.495	0.909	irs	0.562	0.617	0.91	drs	0.771	0.789	0.977	irs
Malt liquors & malt	0.197	1	0.197	drs	0.781	1	0.781	drs	0.677	1	0.677	drs
												(Continued)

Table3. (Cor	ntinued)											
Sub-		20:	15			20	16			201	17	
sectors	ΤE	PTE	SE	TFS	ΤE	PTE	SE	TFS	TE	PTE	SE	TFS
Soft drinks & mineral water	0.59	0.741	0.796	drs	0.135	0.673	0.2	drs	0.119	0.307	0.387	drs
Tobacco products	0.648	0.658	0.986	irs	1		1	1	1	1	-	I
Spinning, weaving & textiles	60.0	0.226	0.398	drs	0.071	0.274	0.259	drs	0.11	0.217	0.507	drs
Cordage, rope, twine & netting	0.242	0.247	0.981	irs	0.171	0.213	0.8	drs	0.086	0.086	0.995	1
Knitting mills	0.03	1	0.03	irs	0.161	0.185	0.87	drs	0.333	0.641	0.519	irs
Wearing apparel except fur apparel	0.162	0.164	0.992	irs	0.517	0.981	0.527	drs	-	1	7	- 1
Tanning & dressing of leather, luggage & handbags	0.142	0.142	666.0	1	0.195	0.311	0.627	drs	0.898	6.0	0.997	irs
Footwear	0.316	0.317	0.998	irs	0.127	0.239	0.529	drs	0.161	0.161	1	I
Wood & wood products	0.153	0.155	66.0	irs	0.192	0.587	0.327	drs	0.181	0.182	0.995	irs
Paper	0.243	0.245	0.991	irs	0.33	0.372	0.886	drs	0.442	0.443	666.0	irs
Publishing & printing services	0.342	0.343	0.998	irs	0.482	0.888	0.543	drs	0.225	0.652	0.346	drs
												(Continued)

l ables. (cor	ntinued)											
Sub-		20	15			201	.6			201	17	
sectors	TE	PTE	SE	TFS	2	PTE	SE	TFS	H	PTE	SE	TFS
Basic chemicals	0.289	0.3	0.961	irs	0.491	0.564	0.872	drs	0.111	0.117	0.942	irs
Paints, varnishes & mastics	0.907	0.93	0.973	irs	0.309	0.344	0.898	drs	0.61	0.608	0.998	irs
Pharmac euticals & medicinal chemicals	0.172	0.17	0.993	irs	0.425	0.627	0.678	drs	0.18	0.176	0.998	1
Soap and detergents cleaning	0.371	0.37	0.998	drs	0.239	0.564	0.424	drs	0.22	0.34	0.639	drs
Chemical products n. e.c.	0.258	0.27	0.97	irs	0.307	0.456	0.674	drs	0.64	0.666	0.964	irs
Rubber	0.366	0.42	0.883	irs	1	1	1	1	0.89	0.92	0.972	irs
Plastic	0.185	0.3	0.622	drs	0.261	0.971	0.269	drs	0.22	0.621	0.347	drs
Glass	0.623	0.73	0.854	irs	0.226	0.296	0.764	drs	0.31	0.313	0.987	drs
Structural clay	0.132	0.14	0.931	irs	0.084	0.118	0.708	drs	0.18	0.183	0.991	irs
Cement, lime & plaster	0.496	0.87	0.567	drs	0.23	0.594	0.388	drs	0.24	0.755	0.321	drs
Articles of concrete & plaster	0.479	0.67	0.716	drs	0.213	0.662	0.321	drs	0.28	0.466	0.594	drs
Non- metallic mineral	0.151	0.15	0.992	irs	0.083	0.163	0.511	drs	0.14	0.143	666.0	1
												(Continued)

Erena et al., Cogent Economics & Finance (2021), 9: 1997160 https://doi.org/10.1080/23322039.2021.1997160

Table3. (Col	tinued)											
Sub-		20	15			20	16			201	1	
sectors	ΤE	PTE	SE	TFS	TE	PTE	SE	TFS	Ξ	PTE	SE	TFS
Basic iron & steel	0.417	0.61	0.679	drs	0.291	0.99	0.294	drs	0.32	0.714	0.45	drs
Structural metal	0.414	0.65	0.635	drs	0.174	0.705	0.247	drs	0.28	0.407	0.697	drs
Cutlery, hand tools & general hardware	0.35	-	0.35	irs	0.153	0.158	0.97	drs	0.22	0.226	0.961	irs
Other fabricated metal	0.622	0.74	0.843	irs	0.225	0.239	0.943	drs	0.03	0.03	0.963	irs
Other general- purpose machinery	0.518	0.6	0.864	irs	0.269	0.499	0.54	drs	0.29	0.309	0.938	Si
Parts & accessories for motor vehicles	1	۲	~ -1	1	0.273	0.67	0.408	drs	0.34	0.533	0.645	drs
Passenger cars, comm ercial vehicles & busses	0.042	0.05	0.87	i.c	1	1	1	1	0.44	1	0.437	Irs
Furniture	0.156	0.29	0.532	drs	0.197	0.471	0.419	drs	0.3	0.874	0.345	drs
Mean	0.347	97.0	0.837		0.325	0.531	0.629		0.37	0.492	0.784	
Note: TE- total to scale;—Dasł	technical efficie - constant retu	ncy computed b rrn to scale.	y CRS model; P1	rE- pure technica	l efficiency com	puted by VRS mo	odel; SE- scale e	efficiency; TFS- ty	pe of scale; drs-	decreasing returi	n to scale; irs-ir	creasing return

Table 4.	Average	annual te	chnical et	fficiency					
Туре	2010	2011	2012	2013	2014	2015	2016	2017	AV
TE	0.343	0.34	0.51	0.385	0.361	0.347	0.325	0.37	0.37
PTE	0.409	0.44	0.70	0.523	0.524	0.455	0.531	0.49	0.51
SE	0.869	0.81	0.75	0.757	0.715	0.837	0.629	0.78	0.76

Where: AV - overall average; TE- technical efficiency; PTE- pure technical efficiency; SE -scale efficiency.

investing in capital expenditure. This result is consistent with prior findings (Abdi, 2008; Gort et al., 1999) that capital expenditures play a crucial role in long-term growth. For instance, Abdi (2008) pointed out investments in both machinery and equipment and non-machinery & equipment have a direct and significant impact on productivity and output levels.

"Insert Table 6 here please"

The account book ratio, which is used in this study as a proxy for internal control appeared positive and has a significant influence on technical efficiency. It indicates an effective internal control system helps firms reduce resource wastage and improves productivity. This finding seems new as it is ignored in prior studies so that one can further investigate more broadly on the role of internal control in enhancing firm technical efficiency. This result is consistent with the finding of Bekele and Belay (2007) that indicated firms that keep a complete book of account are in a better position to prudently plan and track the day-to-day operations of their production unit.

The sign of capital intensity is positive but statistically weak (significant at 10%) impact on efficiency. It slightly supports that capital-intensive industries are more likely to be efficient than labor-intensive industries. Similarly, Alvarez and Crespi (2003) suggest firms that are capital-oriented tend to be more efficient.

4.6. The Malmquist index summary of industries results

We compute Malmquist productivity index output-oriented using panel data (2010 to 2017) of 43 sub-sectors, in which the year 2010 is the base period. The Malmquist index for each year is computed. However, we present the Malmquist index industry mean of the observation periods and Malmquist index annual mean for clarity.

Table 7 presents the Malmquist index summary of sub-sectors geometric means, which contain technical efficiency change, technological change, PTE change, SE change, and total productivity growth. According to the technical efficiency change index, about 21 (48.8%) sub-sectors have experienced technical efficiency progress. Among the sub-sectors which made progress, knitting from the textile industrial group recorded the highest growth (49.9%) followed by wearing apparel (42.4%). In contrast, 22 sub-sectors experienced a decline in technical efficiency. Other fabricated metal products sub-sector faced the largest deterioration (31%) in technical efficiency. The technical efficiency of the tobacco product sub-sector remained unchanged. Moreover, the sector has shown TE progress by 2.2% from 2010 to 2017. The result is consistent with the findings by Abegaz (2013) that the share of technical efficiency change in TFP growth is small and negative in most of the industrial groups.

When looking at the technological index, 40 (93%) sub-sectors experienced technological progress whereas, the remaining 3 (7%) sub-sectors (bakery products; cordage, rope, twine & netting; and structural clay products) recorded regress over the study periods. This means 40 sub-sectors were able to cause shifts in their frontier due to innovation. The top-five in technological progress sub-sectors are cement, lime, and plaster (37%); other general-purpose machinery (34.5%); malt liquors & malt (32%); structural metal products, tanks & containers of metal (24.1%); and other

Table 5. TE, PTE, 9	SE, and TFS-publid	c and private sub-	-sectors					
Major sub-sectors	2015							
1	Public				Private			
1	TE	PTE	SE	TFS	TE	PTE	SE	TFS
Food products and beverages	1.000	1.000	1.000		0.950	1.000	0.950	drs
Textiles	0.309	0.311	0.991	irs	0.697	0.721	0.967	drs
Wearing apparel, except fur apparel	0.374	0.759	0.493	irs	1.000	1.000	1.000	
Tanning of leather, footwear & luggage	0.481	1.000	0.481	irs	0.705	0.834	0.845	drs
Wood, wood products & cork	0.609	1.000	0.609	irs	0.645	1.000	0.645	irs
Paper, paper products and printing	0.493	0.509	0.968	irs	0.794	0.795	0.998	drs
Chemicals and chemical products	0.379	0.448	0.846	irs	0.728	1.000	0.728	drs
Rubber and plastic products	0.642	0.684	0.940	irs	0.566	0.789	0.717	drs
Other non- metallic mineral products	1.000	1.000	1.000	1	1.000	1.000	1.000	I
Basic iron and steel	0.765	0.779	0.982	irs	1.000	1.000	1.000	-
Fabricated metal products	0.409	0.424	0.963	irs	1.000	1.000	1.000	-
Machinery and equipment n.e.c.	1.000	1.000	1.000	1	0.779	1.000	0.779	irs
								(Continued)

Erena et al., Cogent Economics & Finance (2021), 9: 1997160 https://doi.org/10.1080/23322039.2021.1997160

Table5. (Continue	6							
Motor vehicles, trailers & semi- trailers	1.000	1.000	1.000	I	1.000	1.000	1.000	1
Furniture; manufacturing n. e.c.	0.378	0.394	0.959	irs	0.631	1.000	0.631	drs
Mean	0.631	0.736	0.874		0.821	0.939	0.876	
Major sub-sectors	2017							
I	Public				Private			
1	TE	PTE	SE	TFS	Ш	PTE	SE	TFS
Food products and beverages	0.140	0.794	0.177	drs	0.961	1.000	0.961	drs
Textiles	0.032	0.075	0.422	drs	0.551	0.564	0.976	irs
Wearing apparel, except fur apparel	1.000	1.000	1.000	1	0.627	0.717	0.875	irs
Tanning of leather, footwear & luggage	1.000	1.000	1.000	1	0.530	0.540	0.982	irs
Wood, wood products & cork	0.095	0.095	0.994	drs	0.519	1.000	0.519	irs
Paper, paper products and printing	0.088	0.129	0.678	drs	1.000	1.000	1.000	-
Chemicals and chemical products	0.010	1.000	0.010	irs	0.857	0.970	0.883	drs
Rubber and plastic products	0.369	0.411	0.896	drs	0.665	0.931	0.715	drs
Other non- metallic mineral products	0.337	0.442	0.761	drs	0.988	1.000	0.988	drs
								(Continued)

Table5. (Continu	ed)							
Basic iron and steel	0.062	0.078	267.0	irs	1.000	1.000	1.000	1
Fabricated metal products	0.056	0.056	266.0	1	0.881	0.961	0.917	drs
Machinery and equipment n.e.c.	1.000	1.000	1.000	1	0.481	0.744	0.646	irs
Motor vehicles, trailers & semi- trailers	0.178	0.334	0.531	drs	1.000	1.000	1.000	1
Furniture; manufacturing n. e.c.	0.187	0.188	0.995	irs	1.000	1.000	1.000	1
Mean	0.325	0.472	0.733		0.790	0.888	0.890	
Note: TE- total techn return to scale;—Das	ical efficiency compute h- constant return to s	ed by CRS model; PTE- F scale.	oure technical efficien	cy computed by VRS mo	del; SE- scale efficienc	y; TFS- type of scale; d	rs-decreasing returr	n to scale; irs-increasing

asing	
s-incre	
ale; irs	
to sco	
eturn	
ing re	
creas	
Irs-de	
cale; c	
of so	
- typ∈	
r, TFS	
ciency	
e effi	
:- sca	
lel; SE	
S moc	
by VR.	
uted	
comp	
ency	
effici	
hnical	
e tecl	
Ind -3	
el; PT	
pom :	
y CRS	e.
uted b	o scal
ompi	turn t
ency (int rei
effici	constc
nical	ash- c
nl tech	e;-D
- totc	o scal
e: TE	urn ti

Table 6. Tobit and	lysis of the detern	ninants of technica	l efficiency	
	Dependent vo	ıriable: Technical et	fficiency score	
Variables	Coefficient	Standard Error	Z-Value	P-Value
Advertising ratio	-0.0033807	0.0041182	-0.82	0.412
Account book ratio	0.3028605	0.1305197	2.32	0.020**
Skill intensity	-0.0526141	0.1077387	-0.49	0.625
Industry size	-0.029757	0.1191436	-0.25	0.803
Capital expenditure ratio	0.2872345	0.0883343	3.25	0.001***
Capital intensity	0.0000537	0.0000321	1.67	0.094*
Constant	0.0813547	0.1075105	0.76	0.449

Observation: 114 Wald chi2(6) = 23.27 P-value = 0.0007

Note: ***, **, * represent significant level at 1%, 5% & 10% respectively. Advertising ratio—advertising expense/total expenses; Account book ratio—firms which have a book of account/total number of firms in the industry; Skill intensity —production workers/total employees; Industry size—firms with 50 and over employees/number of firms in the industry; Capital expenditure ratio—annual capital expenditure/total fixed assets; Capital intensity—machinery & equipment/production workers.

fabricated metal products (17.2%). The largest deterioration in technical change has been witnessed in the cordage, rope, twine and netting sub-sector (5.5%). On average, the sector has shown 10.5% progress in the frontier shift (innovation).

The overall productivity growth, which is reflected by the Malmquist index, shows 33 sub-sectors had experienced growth in productivity. The first five sub-sectors that experienced growth in productivity are knitting mills (61.6%); wearing apparel (51.6%); rubber products (50.3%); tanning & dressing of leather, luggage & handbag (46.7%); and malt liquors & malt (42.8%). Among the sub-sectors with negative productivity growth, other fabricated metal products sub-sector (19.5%) faced the highest decline in TFP by 19.5%, which is contributed by a decline in technical efficiency change. The negative production growth suggests that the firm produced less output per unit of resources consumed in 2010 compared with 2017. The deterioration of technical efficiency was the major factor for the negative production growth of the industries. The change in TFP has been observed highly because of technological progress. This can be witnessed that several new and improved products are coming to market currently, suggesting an increase in product and process innovation in the sector. Gebreeyesus (2007) found consistent results that the Ethiopian manufacturing sector exhibited an annual average productivity growth of about 9.3 percent between 1996 and 2003. There was a large variation in productivity growth among the industries. On average, the TE, technological change, and productivity showed growth of 2.2%, 10.5%, and 13%, respectively over 2010 to 2017.

Table 8 shows the mean efficiency growth rates for the sub-sectors over various periods (2010–2017) to investigate the trend of the change in the efficiency measures. The progress in PTE, which is one of the important elements of the TE index, contributed to TE improvement. In terms of TE, 2012 is the year when the highest progress was made while 2013 is the year when the heaviest retrogression was observed. In terms of technological progress, 2016 witnessed the highest increase while 2015 showed the lowest. Accordingly, the highest rate in TFP was reached in 2016 whereas the lowest rate was recorded in 2015. In other words, 2016 was also characterized by the year in which the highest economic growth (GDP of 10.9%) has been recorded in Ethiopia.

The year 2011 is characterized by an increase both in technical efficiency and technological progress. The industry's improved utilization of resources and innovation caused an increase in productivity by 27.7% in the year. However, 2012, 2015, and 2017 are the years in which

Table 7. The Malmquist index su	ummary of industries m	eans (2010–2017)			
Sun-sectors	EFFCH	TECHCH	PTECH	SECH	TFPCH
Processing & preserving of meat, fruit & vegetables	0.92	1.106	0.937	0.982	1.018
Vegetable & animal oils and fats	1.251	1.114	1.248	1.003	1.393
Dairy	1.094	1.118	1.095	0.999	1.223
Grain mill	1.1	1.143	1.133	0.972	1.258
Animal feeds	1.187	1.054	1.194	0.995	1.251
Bakery	1.026	0.965	0.978	1.049	0.99
Sugar and sugar confectionery	0.91	1.038	0.91	1	0.945
Macaroni and spaghetti	0.962	1.164	1.009	0.954	1.12
Food products n.e.c.	0.986	1.055	1.078	0.915	1.04
Distilling, rectifying & spirits	0.874	1.111	0.886	0.987	0.971
Wines	1.005	1.12	0.997	1.008	1.125
Malt liquors & malt	1.082	1.32	1.132	0.956	1.428
Soft drinks & mineral water	0.825	1.093	0.944	0.874	0.902
Tobacco products	1	1.165	1	1	1.165
Spinning, weaving & textiles	0.974	1.098	0.958	1.016	1.069
Cordage, rope, twine & netting	0.919	0.945	0.834	1.102	0.868
Knitting mills	1.499	1.078	0.938	1.597	1.616
Wearing apparel except fur apparel	1.424	1.065	1.41	1.01	1.516
Tanning & dressing of leather, luggage & handbags	1.349	1.087	1.349	1	1.467
Footwear	1.043	1.037	1.043	7	1.082
Wood & wood products	1.265	1.049	1.265	1	1.327
					(Continued)

Table7. (Continued)					
Sun-sectors	EFFCH	TECHCH	PTECH	SECH	TFPCH
Paper	1.084	1.107	1.084	1	1.2
Publishing & printing services	0.979	1.102	1.11	0.882	1.079
Basic chemicals	0.883	1.052	0.89	0.992	0.93
Paints, varnishes & mastics	0.947	1.055	0.943	1.004	0.999
Pharmaceuticals & medicinal chemicals	0.924	1.116	0.923	1.001	1.031
Soap and detergents cleaning	0.942	1.107	666.0	0.942	1.042
Chemical products n.e.c.	0.939	1.028	0.944	0.995	0.965
Rubber	1.304	1.153	1.304	1	1.503
Plastic	0.962	1.109	1.06	0.908	1.068
Glass	1.057	1.143	1.052	1.005	1.208
Structural clay	1.083	0.975	1.074	1.008	1.056
Cement, lime & plaster	0.817	1.37	0.961	0.85	1.119
Articles of concrete $\&$ plaster	1.163	1.119	1.181	0.984	1.302
Non-metallic mineral	1.07	1.062	1.04	1.029	1.137
Basic iron & steel	1.021	1.156	1.143	0.893	1.18
Structural metal	0.98	1.241	1.031	0.95	1.216
Cutlery, hand tools & general hardware	0.931	1.098	0.93	1.001	1.023
Other fabricated metal	0.687	1.172	0.69	0.996	0.805
Other general-purpose machinery	1.004	1.345	1.01	0.99 4	1.35
Parts & accessories for motor vehicles	0.911	1.126	0.968	0.942	1.027
Passenger cars, commercial vehicles & busses	1.077	1.097	1	1.077	1.181
					(Continued)

Table7. (Continued)					
Sun-sectors	EFFCH	ТЕСНСН	PTECH	SECH	TFPCH
Furniture	0.997	1.001	1.095	0.911	0.999
Mean	1.022	1.105	1.032	0.991	1.13
		"I mine tochnical afficiance of a		TTDC11 total factor and durationity of	

Where: EFFCH-effficiency change; TECHCH- technology change; PTECH- pure technical efficiency change; SECH—scale efficiency change; TFPCH-total factor productivity change.

technological retrogression was shown which in turn caused negative productivity growth. The technical efficiency index shows worsening from 2013 to 2016. Generally, the industries are more focused on technological progress or innovation than technical efficiency during the study period.

5. Conclusions

This study estimates technical efficiency and total productivity growth of medium and large-scale manufacturing sub-sectors using census data annually collected by the Ethiopian Central Statistical Agency. A technical efficiency score is computed by a data envelopment analysis, whereas the Malmquist productivity index has been employed to estimate total productivity change. Furthermore, a censored Tobit regression model was used to identify potential factors which can define the variation in total technical efficiency scores.

The result shows that on average, medium- and large-scale manufacturing sub-sectors registered a 0.37 (37%) efficiency score over the study periods 2010 to 2017. It suggests that on average, the sector could minimize its input quantity by 63% without altering the level of production or could produce about 63% of production from the resources assumed in the observation periods. In practice, the sector has been suffering from a lack of adequate materials, electric power interruption, and skilled labor. It also could not appropriately utilize the available resources. Thus, it can be concluded that the manufacturing sector should look into its resource utilization methods to obtain the optimal benefit. The Malmquist productivity index shows on average, the sub-sectors made technological progress by 10.5%. The technological change index positive value indicates a decline in the quantity of output produced by a similar quantity of input. It indicates progress in innovation, which has greatly contributed to positive productivity growth in 31 sub-sectors. It suggests that most sub-sectors have paid more focus on technological change than technical efficiency change. Moreover, productivity grew by 13% over the study periods, 2010 to 2017, which is less than 2% per annum. As productivity is the linear combination of catch-up and frontier shift, firms need to balance these factors in order to improve productivity.

Furthermore, we observed that the total technical efficiency scores computed by a constant return to scale model show considerable variations among the sub-sectors under consideration. To understand the determinant factors that can cause a firm to be more efficient or less efficient, a censored Tobit regression was run and the results showed that capital expenditure ratio and account book ratio has a significant positive effect on technical efficiency. The capital expenditure ratio indicates long-term investments that can increase a firm's future cash flow, which in turn improves technical efficiency. On the other hand, the account book ratio reflects firm internal control practice which involves financial management practices and how business transactions are recorded, maintained, and processed into information helpful to decision-makers in planning, directing, and controlling activities. It can be inferred that a firm that designs an effective

Table 8. The M	almquist index s	summary of ann	ual means		
Year	TE	TeChE	PTE	SE	TFP
2011	1.062	1.201	1.12	0.948	1.276
2012	1.544	0.869	1.767	0.873	1.341
2013	0.744	1.225	0.68	1.094	0.911
2014	0.957	1.354	1.036	0.925	1.296
2015	0.948	0.873	0.827	1.147	0.828
2016	0.931	1.46	1.236	0.754	1.36
2017	1.132	0.912	0.876	1.292	1.033
Mean	1.022	1.105	1.032	0.991	1.13

Note: All Malmquist indexes represent geometric means (Coelli, 1996). TE-total technical efficiency; TeChE-technological change; PTE-pure technical efficiency; SE- scale efficiency; TFP-total factor productivity.

accounting system and financial management practices more likely to be efficient. The coefficient of capital intensity is positive but not statically strong. This study also finds public-owned sub-sectors are less efficient than privately owned sub-sectors.

Even though the sector shows total productivity progress, it is still insignificant when compared to other industries. For example, the sector's net contribution to GDP in 2016/17 was 1.1%, while agriculture and services accounted for 36.3% and 39.3%, respectively (CSA, 2018). Growth in the manufacturing sector is expected to be a positive function of the GDP. To meet these expectations, the sector needs to be efficient. Efficiency in the sector can be improved by strengthening the corporate governance structure (Bris et al., 2008), enhancing R&D, innovations, and utilization of information technology. Finally, human resource development is another aspect that needs to be addressed. The skills of personnel working in manufacturing should match the changing requirements of these industries, which are forced upon us by globalization. Without a competent workforce, it is difficult to compete, particularly in this type of knowledge-based industry.

6. Practical implications of the study

The findings of the study would have implications for policymakers and firm owners in that it offers an insight into the source of productivity growth, competitiveness, and areas for further improving the manufacturing sector. Policymakers would also use the findings in designing strategic plans towards increasing the productivity of the sector. The main source of productivity is internal factors which involve optimal usage of existing resources or producing the optimal production from the existing input resources. Thus, considerable due attention should be given to the productivitydriven growth strategy than the foreign direct investment-driven growth strategy of the sector.

7. Limitations and future research directions

Our study has two potential limitations. First, our analysis focuses mainly on medium and largescale firms, excluding small manufacturing firms that make up a large percentage of the sector. Future research may focus on small and medium enterprises (SME) in developing countries like Ethiopia. Second, the limitations in the data also forced us to use sectoral data instead of firm data, which would have allowed a deeper and more interesting or novel analysis. There are several factors recommended in the empirical literature (Alvarez & Crespi, 2003) like export performance, firm owner education level, import performance, research & development, but we could not include these variables in the model due to data unavailability.

Acknowledgements

The authors are thankful to the Office of the Vice President for Research and Technology Transfer of Hawassa University and Internal Grant Agency of the Faculty of Management and Economics, Tomas Bata University in Zlin (Grant Number: IGA/FaME/2020/003) for financial support towards carrying out this research. The authors also would like to thank Prof. Christian Nsiah and two anonymous reviewers for their time and effort devoted to critical review, helpful and constructive comments throughout the revision process.

Funding

This work was supported by the Hawassa University and Tomas Bata University in Zlin (IGA/FaME/2020/003).

Author details

Obsa Teferi Erena¹ E-mail: obteferi@gmail.com ORCID ID: http://orcid.org/0000-0003-4304-5359 Mesfin Mala Kalko² E-mail: kalko@utb.cz ORCID ID: http://orcid.org/0000-0001-5153-4764 Sara Adugna Debele¹ E-mail: saraadu2@gmail.com ORCID ID: http://orcid.org/0000-0002-9832-7049

- ¹ College of Business and Economics, Hawassa University, Hawassa, Ethiopia.
- ² Faculty of Management and Economics, Tomas Bata University in Zlin, Zlin, Czech Republic.

Disclosure statement

No potential conflict of interest was reported by the author(s).

Citation information

Cite this article as: Technical efficiency, technological progress and productivity growth of large and medium manufacturing industries in Ethiopia: A data envelopment analysis, Obsa Teferi Erena, Mesfin Mala Kalko & Sara Adugna Debele, *Cogent Economics & Finance* (2021), 9: 1997160.

References

- Abbott, M., & Doucouliagos, C. (2003). The efficiency of Australian universities: A data envelopment analysis. Economics of Education Review, 22(1), 89–97. https:// doi.org/10.1016/s0272-7757(01)00068-1
- Abdi, T. (2008). Machinery & equipment investment and growth: Evidence from the Canadian manufacturing sector. Applied Economics, 40(4), 465 478. https://doi. org/doi:10.1080/00036840600690215

- Abegaz, M., (2013). Total factor productivity and technical efficiency in the Ethiopian manufacturing sector. EDRI Working Paper 10. Ethiopian Development Research Institute, Addis Ababa, Ethiopia
- Abenoja, Z. R., & Lapid, D. (1991). Barriers to entry, market concentration, and wages in the Philippine manufacturing sector. *Philippine Review of Economics and Business*, 28(2), 191–217. https://pre.econ.upd.edu. ph/index.php/pre/article/view/259/447
- Abramovitz, M. (1956). Resource and output trends in the United States since 1870. American Economic Growth, 46, 5–23. http://www.nber.org/chapters/ c5650
- Acemoglu, D., & Zilibotti, F. (2001). Productivity differences. The Quarterly Journal of Economics, 116 (2), 563–606. https://doi.org/10.1162/ 00335530151144104
- Aigner, D., Lovell, K., & Schmidt, P. (1977). Formulation and estimation of stochastic frontier production function models. *Journal of Econometrics*, 6(1), 21–37. https:// doi.org/10.1016/0304-4076(77)90052-5
- Alvarez, R., & Crespi, G. (2003). Determinants of technical efficiency in small firms. *Small Business Economics*, 20(3), 233–244. https://doi.org/10.1023/ a:1022804419183
- Arrow, K. J., Chenery, H. B., Minhas, B. S., & Solow, R. M. (1961). Capital-labor substitution and economic efficiency. *The Review of Economics and Statistics*, 43 (3), 225. https://doi.org/10.2307/1927286
- Ayelign, Y., & Singh, L. (2019). Comparison of recent developments in productivity estimation: Application on Ethiopian manufacturing sector. Academic Journal of Economic Studies, 5(3), 20–31. https://www.ceeol. com/search/article-detail?id=795796
- Balk, M. B. (2001). Scale efficiency and productivity change. Journal of Productivity Analysis, 15(3), 159–183. https://doi.org/10.1023/a:1011117324278
- Banker, R. D., Charnes, A., & Cooper, W. W. (1984). Some models for estimating technical and scale inefficiencies in data envelopment analysis. *Management Science*, 30(9), 1078–1092. https://doi.org/10.1287/ mnsc.30.9.1078
- Befekadu, D., & Berhanu, N. (2000). Annual report on the Ethiopian economy. *Ethiopian Economic Association*, 1 (2), 1999/2000. Addis Ababa, Ethiopia
- Bekele, T., & Belay, K. (2007). Technical efficiency of the Ethiopia grain mill products manufacturing industry. Journal of Rural Development, 29(6), 45–65. https:// repository.krei.re.kr/bitstream/2018.oak/18653/1/ Technical%20efficiency%20of%20the%20Ethiopian %20grain%20mill%20products%20manufacturing% 20industry.pdf
- Bernard, A. B., & Jensen, B. J. (1999). Exceptional exporter performance: Cause, effect, or both?. Journal of International Economics, 47(1), 1–25. https://doi.org/ 10.1016/s0022-1996(98)00027-0
- Betancourt, R. R., & Clague, C. K. (1975). An economic analysis of capital utilization. Southern Economic Journal, 42(1), 69–78. https://doi.org/10.2307/ 1056564
- Bhat, R., Verma, B. B., & Reuben, E. (2001). Hospital efficiency: An empirical analysis of district hospitals and grant-in-aid hospitals in Gujarat. *Journal of Health Management*, 3(2), 167–197. https://doi.org/10.1177/ 097206340100300202
- Bhatia, D. P. (1990). Misleading growth rates in the manufacturing sector of India. The Journal of Income and Wealth, 12, 222–225.
- Blomström, M., & Persson, H. (1983). Foreign investment and spillover efficiency in an underdeveloped economy: Evidence from the Mexican manufacturing

industry. World Development, 11(6), 493-501. https:// doi.org/10.1016/0305-750x(83)90016-5

- Bris, A., Brisley, N., & Cabolis, C. (2008). Adopting better corporate governance: Evidence from cross-border mergers. *Journal of Corporate Finance*, 14(3), 224–240. https://doi.org/10.1016/j.jcorpfin.2008.03.005
- Caves, D. W., Christensen, L. R., & Diewert, W. E. (1982). The economic theory of index numbers and the measurement of input, output, and productivity. *Econometrica*, 50(6), 1393–1414. https://doi.org/10. 2307/1913388
- Charnes, A., Cooper, W. W., & Rhodes, E. (1978). Measuring the efficiency of decision-making units. European Journal of Operations Research, 2(6), 429–444. https://doi.org/10.1016/0377-2217(78) 90138-8
- Coelli, T. (1996). A guide to DEAP version 2.1, A data envelopment analysis (computer) program, CEPA Working Paper 96/08, University of New England, Australia.
- Comanor, W. S., & Wilson, T. A. (1967). Advertising, market structure and performance. *Review of Economics* and Statistics, 49(4), 423–440. https://doi.org/10. 2307/1928327
- Cooper, W. W., Seiford, L., & Zhu, J. (2004). Handbook of DEA. Kluwer Academic Publishers.
- Cooper, W. W., Seiford, L. M., & Tone, K. (2006). Introduction to data envelopment analysis and its uses. In With DEA-Solver software and references. Springer. 125.
- CSA. (2018). Report on large and medium scale manufacturing and electricity industries survey. The FDRE Statistical Bulletin.
- Delong, J. B. D., & Summers, L. H. (1991). Equipment investment and economic growth. *The Quarterly Journal of Economics*, 106(2), 445–502. https://doi. org/10.2307/2937944
- Denison, E. F. (1962). United States economic growth. The Journal of Business, 35(2), 109–121. https://doi.org/ 10.1086/294483
- Diaz, M. A., & Sanchez, R. (2007). Firm size and productivity in Spain: A stochastic frontier analysis. Small Business Economics, 30(3), 315–323. https://doi.org/ 10.1007/s11187-007-9058-x
- Diewert, W. E. (2000). The challenge of total factor productivity measurement. *International Productivity Monitor*, 1, 45–52. http://econ2.econ.iastate.edu/tes fatsi/TotalFactorProd.Diewert.pdf
- Dinh, H., Palmade, V., Chandra, V., & Cossar, F. (2012). Light manufacturing in Africa: Targeted policies to enhance private investment and create jobs. World Bank.
- Färe, R., Grosskopf, S., Lindgren, B., & Roos, P. (1994). Productivity developments in Swedish hospitals: A Malmquist output index approach. Data Envelopment Analysis: Theory, Methodology, and Applications, 253–272. https://doi.org/10.1007/978-94-011-0637-5 13
- Farrell, M. J. (1957). The measurement of productive efficiency. *Journal of the Royal Statistical Society. Series A (General)*, 120(3), 253. https://doi.org/10. 2307/2343100
- Fu, X. (2005). Exports, technical progress and productivity growth in a transition economy: A non-parametric approach for China. *Applied Economics*, 37(7), 725–739. https://doi.org/10.1080/00036840500049041
- Gebreeyesus, M. (2007). Firm turnover and productivity differentials in the Ethiopian manufacturing. *Journal* of Productivity Analysis, 29(2), 113–129. https://doi. org/10.1007/s11123-007-0076-0
- Geroski, P., Machin, S., & Van Reenen, J. (1993). The profitability of innovating firms. *The RAND Journal of*

Economics, 24(2), 198-211. https://doi.org/10.2307/ 2555757

- Geroski, P. A. (1995). What do we know about entry?. International Journal of Industrial Organization, 13(4), 421–440. https://doi.org/10.1016/0167-7187(95) 00498-x
- Gort, M., Greenwood, J., & Rupert, P. (1999). Measuring the rate of technological progress in structures. Review of Economic Dynamics, 2(1), 207–230. https://doi.org/ 10.1006/redy.1998.0046
- Goshu, Y. Y., Kitaw, D., & Matebu, A. (2017). Development of productivity measurement and analysis framework for manufacturing companies. *Journal of Optimization in Industrial Engineering*, 10(22), 1–13. DOI: 10.22094/JOIE.2017.274
- Green, A., & Mayes, D. (1991). Technical inefficiency in manufacturing industries. *The Economic Journal*, 101 (406), 523–538. https://doi.org/10.2307/2233557
- Griliches, Z. (1998). Productivity, R&D, and the data constraint. In Z. Griliches (Ed.), R&D and productivity: The econometric evidence (pp. 347–374). University of Chicago Press. http://www.nber.org/chapters/c8352
- Gumbau-Albert, M., & Maudos, J. (2002). The determinants of efficiency: The case of the Spanish industry. *Applied Economics*, 34(15), 1941–1948. https://doi. org/10.1080/00036840210127213
- Hailu, K. B., & Tanaka, M. (2015). A "true" random effects stochastic frontier analysis for technical efficiency and heterogeneity: Evidence from manufacturing firms in Ethiopia. *Economic Modelling*, 50, 179–192. https://doi.org/10.1016/j.econmod.2015.06.015
- Hossain, M. A., & Karunaratne, N. D. (2004). Trade liberalisation and technical efficiency: Evidence from Bangladesh manufacturing industries. Journal of Development Studies, 40(3), 87–114. https://doi.org/ 10.1080/0022038042000213210
- Jovanovic, B. (1982). Selection and the evolution of industry. *Econometrica*, 50(3), 649–670. https://doi. org/10.2307/1912606
- Kendrick, J. W. (1956). Productivity trends: Capital and labor. Review of Economics and Statistics, 38(3), 248–257. https://doi.org/10.2307/1925777
- Kenya Association of Manufacturers. (2018). Manufacturing in Kenya under the 'Big 4 agenda': A sector deep-dive report.
- Kim, S. (2003). Identifying and estimating sources of technical inefficiency in Korean manufacturing industries. *Contemporary Economic Policy*, 21(1), 132–144. https://doi.org/10.1093/cep/21.1.132
- Kirkley, J., Squires, D., & Strand, I. E. (1998). Characterizing managerial skill and technical efficiency in a fishery. *Journal of Productivity Analysis*, 9(2), 145–160. https://doi.org/10.1023/a:1018308617630
- Lakner, S., Brenes-Muñoz, T., & Brümmer, B. (2017). Technical efficiency in Chilean agribusiness industry: A metafrontier approach. Agribusiness, 33(3), 302– 323. https://doi.org/10.1002/agr.21493
- Larossi, G., Mousley, P., & Radwan, I. (2009). An assessment of the investment climate in Nigeria. In The international bank for reconstruction and development. The World Bank. 110.
- Latruffe, L., Balcombe, K., Davidova, S., & Zawalinska, K. (2004). Determinants of technical efficiency of crop and livestock farms in Poland. *Applied Economics*, 36 (12), 1255–1263. https://doi.org/10.1080/ 0003684042000176793
- Lee, J. D., & Heshmati, A. (2009). Introduction productivity, efficiency, and economic growth in the Asia-Pacific region. In J. D. Lee & A. Heshmati (Eds.), Productivity, efficiency, and economic growth in the Asia-Pacific region. Contributions to economics.

Physica-Verlag HD. 1. https://doi.org/10.1007/978-3-7908-2072-0_1

- Lee, L.-F., & Tyler, W. G. (1978). The stochastic frontier production function and average efficiency: An empirical analysis. *Journal of Econometrics*, 7(3), 385–389. https://doi.org/10.1016/0304-4076(78) 90061-1
- Leibenstein, H., & Maital, S. (1992). Empirical estimation and partitioning of X-inefficiency: A data envelopment approach. The American Economic Review, Papers and Proceedings of the Hundred and Fourth Annual Meeting of the American Economic Association, 82(2), 428–433. https://www.jstor.org/ stable/2117439
- Li, Y., & Zhao, Z. (2017). The dynamic impact of intellectual capital on firm value: Evidence from China. *Applied Economics Letters*, 25(1), 19-23. https://doi. org/10.1080/13504851.2017.1290769
- Lucas, R. E. (1988). On the mechanics of economic development. Journal of Monetary Economics, 22(1), 3–42. https://doi.org/10.1016/0304-3932(88)90168-7
- Lundvall, K., & Battese, G. E. (2000). Firm size, age and efficiency: Evidence from Kenyan manufacturing firms. Journal of Development Studies, 36(3), 146–163. https://doi.org/10.1080/ 00220380008422632
- Mankiw, N. G., Romer, D., & Weil, D. N. (1992). A contribution to the empirics of economic growth. *The Quarterly Journal of Economics*, 107(2), 407–437. https://doi.org/10.2307/2118477
- Martin, J., & Page, J. (1983). The impact of subsidies on X-efficiency in LDC industry: Theory and an empirical test. The Review of Economics and Statistics, 65(4), 608–617. https://doi.org/10.2307/1935929
- Mathijs, E., & Vranken, L. (2000). Farm restructuring and efficiency in transition: Evidence from Bulgaria and Hungary, Selected Paper. American Agricultural Economics, 1262. https://doi.org/10.22004/ag.econ. 21886
- McKinsey. (2012). Manufacturing the future: The next era of global growth and innovation,
- MoFED. (2011). Growth and transformation: Five Years (2011-2015) strategic plan of Ethiopia. The Ministry of Finance and Economic Development, Retrieved www. mofed.gov.et
- Murat, Ş., & Federica, S. (2018). A cross-country analysis of total factor productivity using micro-level data. *Central Bank Review*, 18(1), 13–27. https://doi.org/10. 1016/j.cbrev.2018.01.001
- Naudé, W., & Szirmai, A. (2012). The importance of manufacturing in economic development: Past, present and future perspectives. UNU-MERIT Working Papers, 2012–2041. United Nations University.
- Nelson, R. R., & Winter, S. G. (1973). Toward an evolutionary theory of economic capabilities. Papers and proceedings of the Eighty-fifth annual meeting of the American Economic Association. *The American Economic Review*, 63(2), 440–449. https://doi.org/10. 2307/1817107
- OCED. (2014). Perspectives on global development: Boosting productivity to meet the middle-income challenge
- Oqubay, A. (2015). Made in Africa: Industrial policy in Ethiopia. Oxford University Press. Retrieved https:// library.oapen.org/handle/20.500.12657/40152
- Oqubay, A. (2018). The structure and performance of the Ethiopian manufacturing sector. Working Paper Series N° 299, African Development Bank, Abidjan, Côte d'Ivoire.
- Özçelik, E., & Taymaz, E. (2004). Does innovativeness matter for international competitiveness in

developing countries? The case of Turkish manufacturing industries. *Research Policy*, 33(3), 409–424. https://doi.org/10.1016/j.respol.2003.09.011

- Page, J. M., Jr. (1980). Technical efficiency and economic performance: Some evidence from Ghana. Oxford Economic Papers, 32(2), 319–339. https://doi.org/10. 1093/oxfordjournals.oep.a041482
- Rao and Tesfahunegn. (2015). Performance measurement of manufacturing industries in Ethiopia: An analytical study. *Journal of Poverty, Investment and Development, 7,* 42–54.
- Ray, S. C. (1997). Regional variation in productivity growth in Indian manufacturing: A non-parametric analysis. *Journal of Quantitative Economics*, 13(1), 73–94.
- Rezitis, A. N., & Kalantzi, M. A. (2016). Investigating technical efficiency and its determinants by data envelopment analysis: An application in the Greek food and beverages manufacturing industry. Agribusiness, 32(2), 254–271. https://doi.org/10.1002/agr.21432
- Romer, P. M. (1986). Increasing returns and long-run growth. Journal of Political Economy, 94(5), 1002–1037. https://doi.org/10.1086/261420
- Romer, P. M. (1990). Endogenous Technological Change. Journal of Political Economy, 98(5, Part 2), S71–S102. https://doi.org/10.1086/261725
- Sala-i-Martin, X. (1997). I just ran four million regressions. American Economic Review, 87, 178–183. https://doi. org/10.3386/w6252
- Salim, R. A., & Kalirajan, K. (1999). Sources of output growth in Bangladesh food processing industries: A decomposition analysis. *The Developing Economies*, 37(3), 355–374. https://doi.org/10.1111/j.1746-1049. 1999.tb00237.x
- Sarkis, J. (2000). An analysis of the operational efficiency of major airports in the United States. Journal of Operations Management, 18(3), 335–351. https://doi. org/10.1016/s0272-6963(99)00032-7
- Solow, R. M. (1956). A contribution to the theory of economic growth. The Quarterly Journal of Economics, 70 (1), 65–94. https://doi.org/10.2307/1884513
- Sonobe, T., Akoten, J., & Otsuka, K. (2009). An exploration into the successful development of the Leather-Shoe Industry in Ethiopia. *Review of Development Economics*, 13(4), 719–736. https://doi.org/10.1111/j. 1467-9361.2009.00526.x

- Sullivan, A., & Sheffrin, S. (2007). *Economics: Principles in action*. Pearson Prentice Hall.
- Sun, H., Hone, P., & Doucouliago, H. (1999). Economic openness and technical efficiency: A case study of Chinese manufacturing industries. *The Economics of Transition*, 7(3), 615–636. https://doi.org/10.1111/ 1468-0351.00028
- Sur, A., Nandy, A., & Zhang, X. (2018). FDI, technical efficiency and spillovers: Evidence from the Indian automobile industry. *Cogent Economics & Finance*, 6(1), 1. https://doi.org/10.1080/23322039.2018.1460026
- Sutton, J. (1991). Sunk costs and market structure: Price competition, advertising, and the evolution of concentration. The MIT Press.
- Swan, T. W. (1956). Economic growth and capital accumulation. *Economic Record*, 32(2), 334–361. https://doi.org/10.1111/j.1475-4932.1956.tb00434.x

Taymaz, E., & Saatci, G. (1997). Technical change and efficiency in Turkish manufacturing industries. Journal of Productivity Analysis, 8(4), 461–475. https://doi.org/10.1023/a:1007796311574

- Tsegay, G. T., Tigabu, D. G., Girum, A., & Gebrehiwot, A. (2018). Productivity determinants in the manufacturing sector in Ethiopia: Evidence from the textile and garment industries. Ethiopian Development Institute.
- Tzeng, G.-H., & Huang, -J.-J. (2013). Fuzzy multiple objective decision making. Chapman and Hall/CRC Press, Taylor and Francis Group.
- UNCTAD. (2015). Technology and innovation report: Fostering innovation policies for industrial development.
- UNDP. (2017). Understanding African experiences in formulating and implementing plans for emergence: Growing manufacturing industry in Ethiopia, case study.
- United Nations. (2016). Economic Commission for African. Country Profile.
- Whang, U. (2016). Skilled-labor intensity differences across firms, endogenous product quality, and wage inequality. Open Economies Review, 27(2), 251–292. https://doi.org/10.1007/s11079-015-9370-z
- World Bank. (2014), Kenya economic update, anchoring high growth-can manufacturing contribute more?
- Wu, Y. (1993). Scale, factor intensity and efficiency: An empirical study of the Chinese coal industry. Applied Economics, 25(3), 325–334. https://doi.org/10.1080/ 00036849300000039



© 2021 The Author(s). This open access article is distributed under a Creative Commons Attribution (CC-BY) 4.0 license.

You are free to:

Share — copy and redistribute the material in any medium or format. Adapt — remix, transform, and build upon the material for any purpose, even commercially. The licensor cannot revoke these freedoms as long as you follow the license terms. Under the following terms: Attribution — You must give appropriate credit, provide a link to the license, and indicate if changes were made. You may do so in any reasonable manner, but not in any way that suggests the licensor endorses you or your use. No additional restrictions You may not apply legal terms or technological measures that legally restrict others from doing anything the license permits.

Cogent Economics & Finance (ISSN: 2332-2039) is published by Cogent OA, part of Taylor & Francis Group. Publishing with Cogent OA ensures:

- Immediate, universal access to your article on publication
- High visibility and discoverability via the Cogent OA website as well as Taylor & Francis Online
- Download and citation statistics for your article
- Rapid online publication
- Input from, and dialog with, expert editors and editorial boards
- Retention of full copyright of your article
- Guaranteed legacy preservation of your article
- Discounts and waivers for authors in developing regions

Submit your manuscript to a Cogent OA journal at www.CogentOA.com